KWAME NKRUMAH UNIVERSITY OF SCIENCE AND TECHNOLOGY COLLEGE OF HUMANITIES AND SOCIAL SCIENCES FACULTY OF SOCIAL SCIENCES SCHOOL OF BUSINESS

EVALUATING EFFICIENT SUPPLY CHAIN MANAGEMENT ON SUSTAINABLE
SUPPLY CHAIN IN THE COCOA SECTOR: THE MODERATING EFFECT OF TOP

MANAGEMENT SUPPORT

BY
MARTHA DARKWA
(PG9261821)

A RESEARCH THESIS SUBMITTED TO THE INSTITUTE OF DISTANCE

LEARNING, KWAME NKRUMAH UNIVERSITY OF SCIENCE AND TECHNOLOGY

IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE AWARD OF A

MASTER OF SCIENCE DEGREE IN LOGISTICS AND SUPPLY CHAIN

MANAGEMENT

NOVEMBER, 2023

KNUST



DECLARATION

I hereby declare that this work is the result of my own original work and that to the best of my knowledge no part of it has been presented for another degree masters in this university or elsewhere except where due acknowledgment has been made in the text.

| Martha Darkwa (PG9261821) | | |
|---------------------------|----------------|--|
| Student Name and ID No. | Signature | Date |
| Certified by | | |
| Dr Seth Nkrumah | | 1 |
| Supervisor | Signature | Date |
| Certified by | | |
| Prof. David Asamoah | | THE STATE OF THE S |
| Head of Department | Signature | Date |
| A | CKNOWLEDGEMENT | A CONTRACTOR OF THE PARTY OF TH |

I would like to acknowledge and give my warmest thanks to my supervisor Dr. Seth Nkrumah at the Kwame Nkrumah University of Science and Technology, who in spite of his busy schedules

found time to review my work as I made progress and made this work possible. His time, guidance and advice carried me through all the stages of writing this project. I would also like to thank Mr Dennis Tetteh for his brilliant comments, suggestions and sometimes helped me to rewrite portions of this project.

My sincere thanks and gratitude to my my family for their continuous support and understanding throughout my studies and writing of this project. Your prayer for me was what sustained me this far.

Last but not the least, I would like to thank God, for letting me through all the difficulties. I have experienced your guidance day by day. You are the one who let me finish this programme successfully and would forever be grateful to you.

Finally, I wish to thank all those who in one way or another contributed to the completion and success of this project. I will forever be grateful. Thank you all.

DEDICATION

This thesis is dedicated to the Almighty God who has brought me this far, to my Boss, Mr. Emmanuel Osei Sarfo and to my parents for their love and support towards my education, without whose caring and support it would not have been possible.

KNUST



ABSTRACT

Efficient Supply Chain Management (SCM) has presently become a crucial strategy adopted by many firms and organisations to remain competitive within their respective sectors and as well to continue to be profitable and therefore has been identified as a significant phenomenal component of trade which is a crucial aspect of economic development and activities. With supply chain disruptions impeding the flow of products, hampering firms' abilities to produce their goods as well as adversely impacting on the performance of a supply chain, this study aims at examining the effect efficient supply chain management practices have on sustainable supply chain with a moderation effect of top management. The study with the adoption of a quantitative research method, employed a convenience sampling to selected 10 cocoa companies in the Kumasi metropolis of the Ashanti Region and then, a simple random sampling technique to arrive at surveying 167 staff members of the selected companies. Through the utilisation of SPSS, the study revealed that the cocoa companies adopted the following efficient supply chain practices in their operations; information technology, relationship management, and internal integration. Analysis revealed a significant and positive relationship between efficient supply chain practices and sustainable supply chain. Results further indicates that top management has a positive and significant moderation effect on the relationship between efficient supply chain practices and sustainable supply chain. Study revealed that some of the challenges affecting the adoption and implementation of supply chain practices are companies lack professional expertise, and lack frequent Supply Chain Management training among others. The study recommends that in light of the challenges identified, workshops and seminars should be organised for the workers to help enlightened them on the challenges and how they can be mitigated.

TABLE OF CONTENT

Content

| DECLARATION | i |
|---|------|
| ACKNOWLEDGEMENT | i |
| DEDICATION | |
| ABSTRACT | iv |
| TABLE OF CONTENT | iv |
| LIST OF TABLES | vii |
| LIST OF FIGURES | viii |
| CHAPTER ONE: INTRODUCTION | |
| 1.1 Background of the Study | 1 |
| 1.2 Problem Statement | 3 |
| 1.3 Research Objectives | |
| 1.4 Research Questions. | |
| 1.5 Significance of the Study | 6 |
| 1.6 Limitation, Delimitation, and Scope of the Study | |
| 1.7 Brief Methodology | 8 |
| 1.8 Organisation of the Study | 8 |
| CHAPTER TWO: LITERATURE REVIEW | 9 |
| 2.1 Introduction | 9 |
| 2.2 Conceptual Review | |
| 2.2.1 Supply Chain | |
| 2.2.2 Supply Chain Management | . 11 |
| 2.2.4 Efficient Supply Cha <mark>in Management</mark> | . 13 |
| 2.2.4 Sustainable Supply Chain Management | . 23 |
| 2.2.7 Top Management | . 26 |

| 2.2.8 Challenges Affecting Efficient Supply Chain Management | 27 |
|---|-------------------|
| 2.3 Theoretical Review | 28 |
| 2.3.1 Systems Theory | 28 |
| 2.3.2 Contingency Theory | |
| 2.3.3 Resource Based View Theory | 31 |
| 2.4 Empirical Review | 32 |
| 2.5 Conceptual Framework | 35 |
| 2.5.1 Relationship Between Information Technology and Sustainable Supply Chain | 36 |
| 2.5.2 Relationship Between Relationship Management and Sustainable Supply Chain | 37 |
| 2.5.3 Relationship Between Internal Integration and Sustainable Supply Chain | 38 |
| 2.5.4 Moderating Effect of Top Management Commitment | 39 |
| 2.6 <mark>Summary</mark> | <mark> 4</mark> 0 |
| CHAP <mark>TER THREE: RESEARCH METH</mark> ODOLOGY | 41 |
| 3.1 Introduction | |
| 3.2 Research Design | 41 |
| 3.3 Study Population | 42 |
| 3.4 Sampling Technique and Sampling Size | 42 |
| 3.5 Method of Data Collection | 44 |
| 3.6 Method of Data Analysis | 44 |
| 3.7 Relia <mark>bility and Validi</mark> ty | |
| 3.8 Ethical Consideration | 45 |
| CHAPTER FOUR: DAT <mark>A ANALYSIS AND PRESENTATION</mark> | 46 |
| 4.1 Introduction. | 46 |
| | |

| 4.3 Reliability Analysis | 52 |
|--|-----------------|
| 4.4 Adopted Efficient Supply Chain Management Practices | 53 |
| 4.5 Factors Influencing the Adoption of Efficient SCM Practices | 58 |
| 4.6 The Impact of Efficient SCM Practices on Sustainable Supply Chain | 61 |
| 4.7 Challenges Affecting the Adoption and Implementation of ESCM Practices | 64 |
| CHAPTER FIVE: SUMMARY, CONCLUSION AND RECOMMENDATION | N 65 |
| 5.1 Introduction | 65 |
| 5.2 Summary of Findings | |
| 5.3 Conclusion | 67 |
| 5.4 Recommendation | 68 |
| 5.5 Suggestions for Further Research | |
| REFERENCE | <mark>70</mark> |
| RESEARCH QUESTIONNAIRE | 94 |
| | - |
| LIST OF TABLES | |
| Table | Page |
| | / |
| Table 4.1: Age of Respondents | |
| | |
| Table 4.2: Gender of Respondents | 50 |
| | |
| Table 4.2: Gender of Respondents | 52 |
| Table 4.2: Gender of Respondents Table 4.3: Length of Service Table 4.4: Results of Survey Instrument Reliability Test | 52 55 |
| Table 4.2: Gender of Respondents Table 4.3: Length of Service | 52 55 55 |

| Table 4.8: Extent of Adoption of Internal Integration as an ESCM Practice |
|--|
| Table 4.9: Extent of Adoption of Top Management as an ESCM Practice |
| Table 4.10: Sustainable SC as a Factor Influencing ESCM Practices Adoption |
| Table 4.11: Model Estimation of Efficient SCM Practices and Sustainable Supply Chain 64 |
| Table 4.12: Model Estimation of Top Management |
| Table 4.13: Model Estimation of the Moderating Factor of Top Management |
| Table 4.14: Challenges Affecting the Adoption of Efficient SCM Practices |
| LIST OF FIGURES |
| Figure |
| Figure 2.1: Conceptual Framework |
| Figure 4.1: Education Level of Respondents |
| Figure 4.2: Designation of Respondents |
| 53 Figure 4.3: Staff Strength of Companies |
| 54 |
| The state of the s |
| |
| |
| |
| THE ROLL OF THE PARTY OF THE PA |
| THE WO SANE NO BROWNERS |
| JANE |

CHAPTER ONE INTRODUCTION

1.1 Background of the Study

Supply chain management (SCM) signifies a remarkable transformation in the manner in which organizations perceive their own identity. It has observed the generation of worth through the amalgamation and synchronization of supply, demand, and relationships, in order to effectively and lucratively fulfil the requirements of customers, spanning across both the private and public sectors (Esper et al., 2020). Companies worldwide implement supply chain management due to its proven outcomes, which include the reduction of delivery time, enhanced financial performance, increased customer satisfaction, establishment of trust with suppliers, and various other benefits (Mastos et al., 2020). According to Gouda and Saranga (2018), companies often utilize supply chain practices as a means to enhance their operational proficiency. In recent times, Aid Agencies have displayed a growing inclination towards comprehending the influence of supply chain and its associated stakeholders on organizational strategy (Saeed and Kersten, 2019). The success of every corporation has therefore become imperative especially within the remit of globalisation. It has therefore become the case that one of the components for the realisation of organisational success is dependent on the efficiency of the supply chain system. Although there are different supply chain systems which differ from industry to industry, globalisation has led to a functional integration of activities not only between firms but also among industries (Pananond et al., 2020). Thus, there had been tremendous initiatives to improve the applicability of supply chain management practices and consequently the sustainable supply chain because of its keen WJ SANE NO significance (Zaid et al., 2018).

The achievement of effective SCM is widely considered to be heavily reliant on the imperative to dismantle obstacles, not solely within internal departments and business processes, but also among companies throughout the entire supply chain. Additionally, its triumph is linked to the demanding cultivation of a fresh culture grounded in empowerment, ongoing collaborative learning, and perpetual enhancement (Norman and Naslund, 2019).

Cocoa has been for decades the backbone of the economy of Ghana and its major earner in terms of foreign exchange. It has therefore become the most important agricultural export product for the country as it accounts for about 10% of the country's GDP. The cocoa industry further serves as an employment hub employing well over a million people across the country (Alagidede *et al.*, 2018). In this contemporary era of innovation, organisations are keen on leveraging efficient techniques to enhance their business strategies. Supply chain management practices had become one of the most adopted techniques in terms of leveraging on organisational techniques to enhance business efficiency and profitability. The management of the supply chain systems within the cocoa industry has thus been considered one of the critical drivers of competitiveness in the industry as every organisation within the industry form part of the supply chain. More so as it has become a tool adopted by organisations to attain levels of high competitive advantages (Bai *et al.*, 2022).

As organizations are currently endeavouring to ascertain the means to assimilate decisions pertaining to supply chain functions, facilities dispersed across various geographical locations, and temporal considerations, the imperative of factually grounded supply chain management cannot be overstated. The crux of fact-based supply chain management lies in the amalgamation of planning and control, a construct characterized by three pivotal facets. The initial dimension

pertains to the functional amalgamation, encompassing determinations regarding procurement, production, and distribution operations within the organization, as well as between the organization and its suppliers and clientele. The subsequent dimension encompasses the geographical amalgamation of these functions across tangible establishments situated on one or multiple continents. Lastly, the third dimension encompasses the temporal amalgamation of strategic, tactical, and operational determinations within the supply chain (Jena and Ghadge, 2021). Global supply chains in the era of globalization are often defined by an increased reliance on transportation, which has clear ramifications for the environment. Additionally, these chains can foster behaviours that may not always align with social sustainability. For instance, the presence of inexpensive manpower may compel enterprises to delegate operations to emerging nations with restricted supervision of labour safety. These elements are prompting interested parties to contemplate the concept of durability owing to the escalating apprehension regarding domestic and international guidelines, as well as the increasing emphasis placed by ultimate consumers on the ramifications of durability (Malak-Rawlikowska et al., 2019; Di Fazio and Modica, 2018).

1.2 Problem Statement

The efficiency and effectiveness of supply chain management have significant implications for an organization's capacity to fulfil customer demands, maintain its reputation, and achieve overall financial prosperity (Xu et al., 2019). According to Arunachalam et al. (2018) and Papalexi et al. (2020), the inefficiency of supply chain management represents the most significant potential for operational inefficiencies within any given organization. Enterprises that engage in the global trade of goods such as cocoa face the challenging responsibility of both sustaining their supply sources and ensuring the continuous functionality of their supply chain operations. Any issue in

the distribution network can result in financial setbacks for the organization or governing body, particularly when the product in query holds significance at a national level. The COVID-19 pandemic has emphasized the significant importance of effective supply chain management in ensuring the triumph of a business. Furthermore, it is essential for the overall well-being of the global economy. The crisis caused by the pandemic had extensive consequences on a global scale, impacting every stage of the supply chain, ranging from the acquisition of raw materials to the production of finished goods. Consequently, numerous companies have been confronted with the imperative task of developing the necessary resilience to withstand such adversities, thereby presenting a critical challenge for supply chain management (Arunmozhi *et al.*, 2021).

While there had been significant development within the supply chain sector of the cocoa industry in Ghana, there had been increasing concerns with regards to environmental issues such as resource depletion, waste production, and emission generation among others which can be traced to activities as a result of SCM applications (Pyykkö *et al.*, 2021). The processes adopted by organisations within the supply chain industry had been questioned as their application of SCM practices have had direct environmental results through operations such as procurement, production, and distribution among others (Abdallah and Al-Ghwayeen, 2019).

Again, whereas literature is replete with studies on supply chain management practices and its impact on organisational performances (Hong *et al.*, 2018; Basheer *et al.*, 2019; Tarigan *et al.*, 2021), there are very few studies focusing on the efficiency of supply chain management and as well sustainable supply chain (Al-Sheyadi *et al.*, 2019; Cankaya and Sezen, 2018; Foo *et al.*, 2018). With some studies such as Zhu *et al.* (2013), Beske *et al.* 2014, and Nimeh *et al.* 2018, focusing on improving the efficiency of supply chain management practices within the context of

the research scope, findings of these studies have remained mixed and inconclusive. Whereas many of these studies on Ghana are focused on sectors such as the manufacturing sector (Afum *et al.*, 2020; Kusi-Sarpong *et al.*, 2019; Famiyeh and Kwarteng, 2018), banking sector (Omoregie *et al.*, 2019; Barber *et al.*, 2017) and SMEs cutting across all sectors (Kot, 2018; Hong *et al.*, 2018), little studies are found with a focus on the cocoa industry (Amoako *et al.*, 2021; GlaveeGeo *et al.*, 2020).

Once again, researchers have made numerous endeavours to examine the correlation between effective practices in supply chain management and sustainability. Nevertheless, the results have exhibited a diversity among various countries and industries (Suryanto *et al.*, 2018; Li *et al.*, 2021). Also, studies on the relationship in emerging countries are very limited. Another frequently observed and distinctive characteristic of a significant proportion of these investigations is their concentration on a limited quantity of enterprises. Thus, examination conducted by this study will therefore help breach the gap in literature.

1.3 Research Objectives

The main aim of this study is to examine the application of efficient supply chain management practices leading to the attainment of a sustainable supply chain system of the cocoa sector in Ghana. Thus the specific objectives of this study are;

i. To determine the effect of efficient supply chain management on sustainable supply chain practices within the cocoa sector of Ghana. ii. To evaluate the moderating effect of top management on the relationship between supply chain management and sustainable supply chain.

1.4 Research Questions

This study is guided by the following research questions;

- i. What is the impact of efficient supply chain management on sustainable supply chain practices within the cocoa sector of Ghana?
- ii. What is the moderating effect of top management support on the relationship between efficient supply chain management and sustainable supply chain?

1.5 Significance of the Study

Cocoa has served as the fundamental pillar of Ghana's economy for a period exceeding sixty years, furnishing approximately one-third of all proceeds derived from exports. This accounts for a range of 25-30 percent of the overall earnings from exports and contributes approximately 10 percent to the Gross Domestic Product (GDP) (Vigneri and Kolavalli, 2017). In spite of the significant contribution of cocoa to the economic landscape of Ghana, it appears that the critical importance of supply chain integration and its successful implementation within the industry is being undervalued. With various policy initiatives undertaken by Government of Ghana and various stakeholder players within the sector, this study could serve as a policy guide to drive sectorial reforms that results in boosting the efficiency within the supply chain sector of the cocoa industry in Ghana and further creating sustainability of the supply chain.

Again, on the organizational level, the lack of advanced technological developments and ineffective dissemination of information throughout the cocoa supply chain has resulted in decreased levels of productivity and efficacy within the industry. It is against the backdrop of these challenges and their associated consequences that prompted the researchers to undertake this study, with the aim of investigating methods to overcome the obstacles to supply chain

sustainability, thereby enhancing cocoa production in Ghana. It is anticipated that this would facilitate the cocoa industry in surmounting the obstacles related to the integration of the supply chain, attaining a competitive edge, and making well-informed decisions to enhance the performance of their enterprises and subsequently enhance the economy of Ghana. In conclusion, the research will make a valuable contribution to the existing body of knowledge on effective and sustainable management of the supply chain and also provide a foundation for future investigations.

1.6 Limitation, Delimitation, and Scope of the Study

The study's limitations are demonstrated by the extent of data utilized, which is solely focused on the cocoa industry of Ghana. Furthermore, similar to all quantitative studies, the methodological approach represents a limitation in itself, as the inquiry into the reason behind the presence or absence of a relationship can never be definitively addressed. More broadly, the exceedingly restricted timeframe, which amounts to approximately two months, for the creation of this thesis, serves as a constraint in and of itself and consequently hindered the researcher from expanding the extent of the investigation, such as, for example, conducting a comparison between companies from two different countries or employing similar methodologies. It is thus imperative to acknowledge that this study is exclusively confined to the concepts of supply chain management practices and sustainable supply chain systems. Geographically, the study is specifically limited to cocoa industry of the Republic of Ghana. Contextually, the research focuses on issues bordering on supply chain management practices and sustainable supply chain systems.

SANE

1.7 Brief Methodology

The research design employed in order to achieve the objectives of the study was that of a Case Study approach, thus, enabling the researcher to attain an in-depth account of the variables employed in the study and to analyse intensively the diverse phenomena of all the variables. The targeted population for the study is experts individual sector players within the cocoa industry of Ghana.

The study adopted a quantitative research approach to collect the data which is common with case studies. Based on the projected purpose of the study, the study employed a primary data collection method thro Self-administrated questionnaires. Descriptive statistics in the form of frequencies, percentages, and central tendencies were therefore employed for the analysis of this study.

1.8 Organisation of the Study

This study is organised into five key chapters which to a larger extent are interconnected. The five key chapters comprise of chapter one focusing on the introduction of the study, the background, the problem statement, the research objective, the research questions, the significance of the study, the scope of the study limitations as well as the organisation of the study. The second chapter which is dubbed chapter two focuses on the review of relevant journals, articles and related works of the study subject. The third chapter which is also dubbed chapter three focuses on the methodology of the study describing the approach and methods adopted during the study. The fourth chapter which is named chapter four focuses on the analysis and discussion of the data obtained. The last chapter which is dubbed chapter five presents the finding, conclusion and gives recommendations.

CHAPTER TWO LITERATURE REVIEW

2.1 Introduction

This chapter of the study intends to present the conceptual, theoretical, and empirical analysis of literature on Supply Chain, Supply Chain Management, and Sustainable Supply Chain Management Practices. It also discusses Supply Chain Performance, and Resource Capability, Challenges of SCM Practices and the conceptual framework of the study and finally summarises the literature discussed. This details were presented from reviewed existing material of reference books, journals, reports, online resources and other works by researchers in the same field.

2.2 Conceptual Review

2.2.1 Supply Chain

According to Min et al. (2019), a supply chain is a comprehensive framework encompassing individuals, resources, and organizations engaged in the procurement, production, and distribution of goods or services to meet customer demands. Accordingly, supply chain management pertains to the amalgamation of raw materials and their constituent elements into a finalized product or service, which is subsequently conveyed to the ultimate consumer (Dietrich et al., 2021). Supply chain is an ever-changing, interconnected system. Supply chain has since witnessed a vast level of modernisation in the past 30 years. This clearly indicates that supply chains are dynamic entities that undergo transformation and modification in terms of their dimensions, structure, and arrangement, as well as in the manner in which they are organized, governed, and overseen. A multitude of factors can trigger the emergence of novel supply chains, such as the advent of technological advancements like flexible displays (Tsing, 2016; MacCarthy et al., 2016), the rise of a product that is new or market segment, such as intelligent timepieces, cannot be overlooked (De Propris and Bailey, 2021), or new geographical markets such as Africa

(El Baz *et al.*, 2018). Supply chains may also experience a decline or even vanish when the level of demand becomes inadequate to propel the entire chain forward. This phenomenon has indeed been observed in the case of disk storage technologies, which have been overshadowed by more adaptable forms of digital storage (Schniederjans *et al.*, 2020; Perboli *et al.*, 2018).

It is contended that underlying economic factors are ultimately the decisive factors of supply chain parameters, ascertaining configuration, magnitude, and the characteristic of transaction (MacCarthy et al., 2016). However, there may exist additional factors that possess substantial impacts on the structure, functioning, and alignment of a supply chain during its temporal progression. It is not solely the economic and technological stimulants that exert their influence upon supply chains, but also the regulatory frameworks (Herczeg et al., 2018), sustainability agendas (Wilhelm et al., 2016), political factors (Mancheri et al., 2019), and strategic choices (Kano et al., 2020) affect the structure and configuration of supply chains. The configuration of supply networks has been greatly altered due to global sourcing strategies (Min et al., 2019). Organizations have taken proactive measures to restructure their networks in order to pursue a manufacturing and/or marketing strategy that will more effectively cater to their target markets. As a result, a variety of economic, technological, environmental, and strategic factors have the potential to impact the participation of different entities within supply chains, the location of value adding activities, the coordination and management of these activities, as well as the overall development and growth of the supply chains (Gaudenzi et al., 2021).

WU SANE NO

2.2.2 Supply Chain Management

Supply Chain Management (SCM) originated in the 1980s as a novel comprehensive ideology for overseeing the complete flow of merchandise from suppliers to the end user. The fundamental principle of this methodology resided in controlling a sequence of supply as if it were a unified entity, with the primary aim of rectifying the inefficient allocation of stock and resources (Pettit et al., 2019). SCM, as stipulated by the Council of Supply Chain Management Professionals (CSCMP), encompasses the comprehensive planning and administration of all facets encompassed in sourcing and procurement, conversion, and the entirety of logistics management activities. The majority of alternate SCM definitions underscore the imperative integration of the interconnections that subsist among supply chain collaborators (Berthold, 2019). For instance, in the view of Zaid et al. (2018), The discipline of Supply Chain Management (SCM) encompasses the management of the interconnected relationships between organizations. These relationships are characterized by the upstream and downstream linkages between the various processes that contribute to the creation of value. This value is ultimately delivered to the end consumer in the form of products and services. SCM functions as an integrative mechanism, tasked with the crucial responsibility of connecting major business functions and processes within and across organizations. The ultimate aim is to establish a cohesive and high-performing business model. This model encompasses all activities related to logistics management and manufacturing operations (Wiedemann et al., 2020).

Supply chain management (SCM) encompasses the coordination of interconnected commercial operations involved in delivering essential goods or services to final customers within supply chain activities (Giannakis *et al.*, 2019). The process encompasses all operations ranging from the acquisition of unprocessed materials to the execution of work in progress protocols, culminating

in the production of consumable goods or services that cater to the ultimate consumers. Supply Chain Management (SCM) constitutes a network of interconnected

organizations that enhance the value of a convoluted flow of inputs, commencing from their origin and concluding with the delivery of final products or services demanded by the designated endcustomers (Haddouch *et al.*, 2019). Supply chain can be described as a series of interconnected procedures that work together to facilitate the exchange of value, information, and physical resources within and across different organizations (Barykin *et al.*, 2021). The "management philosophy" commences with the supplier's provisions and descends towards the stream of end users (Arredondo and Alfaro Tanco, 2021). Supply Chain Management (SCM) demonstrates a robust association with the operational proficiencies of an organization in terms of cost reduction, promise fulfilment, logistic services, and product design (Lee, 2021). Through the implementation of supply chain integration, a company can attain customer priorities, gain a competitive advantage over rivals, and accommodate the preferences of suppliers (Haiyun *et al.*, 2021).

The idea of supply chain management is mainly focused on the harmonisation process that results in customer satisfaction (Mageto and Luke, 2020). This underscores the significance of integrated patterns in the course of production flux from primary resources to commodities reaching the possession of consumers. Operations that arise all through the course are imperative to guarantee that the flux remains unobstructed without barriers, thus facilitating the information mechanism to occur in a transparent manner without the diminution in any of the connections (Casino *et al.*, 2021). All supply chains fundamentally engage in competition for the acquisition of customers through the provision of various products or services. A pivotal aspect of ensuring efficient supply chain management, which effectively caters to the expanding market, involves integrating suppliers into the overarching strategy of the company. The execution of supply chain

management shall diminish the expenses related to operations that arise throughout the entire chain, guaranteeing the preservation of product excellence, which shall ultimately contribute to bestowing value upon consumers in relation to the accessibility of products and promptness of service. Consequently, supply chain management shall furnish a competitive edge both directly and indirectly, particularly in terms of product value (Dutta *et al.*, 2020).

2.2.4 Efficient Supply Chain Management

The efficient administration of an organization's supply chains has demonstrated efficacy in facilitating expedient and dependable provision of superior products and services at minimal expense (Zhu *et al.*, 2022). In order to accomplish this, it is of utmost significance to conduct an assessment of the overall supply chain. This entails utilizing the collective resources of the supply chain participants in the most optimal manner feasible, to deliver products and services that are both competitive and cost-effective. Nevertheless, the absence of suitable systems for measuring performance has proven to be a significant hindrance in effectively managing supply chains (Gružauskas *et al.*, 2018; Mangla *et al.*, 2018).

Over the course of the previous decade, numerous subjects and approaches pertaining to supply chain research have been identified and thoroughly examined (Tayur *et al.*, 1998). Criteria for optimization in supply chain models have encompassed factors such as cost (Lacomme *et al.*, 2018), inventory levels (Brunaud *et al.*, 2019), profit (Raj *et al.*, 2018), fill rate (Longo *et al.*, 2019), stock out probability (Vledder *et al.*, 2019), product demand variance (Dolgui *et al.*, 2020), and system capacity (Tiwari *et al.*, 2018). Most deterministic and stochastic models focus on individual components of the supply chain system, such as the supply-production, productiondistribution, or inventory-distribution systems. Certain models address strategic

concerns related to supply chains, including the optimal placement of plants and warehouses and the flow of goods. In contrast, other models address operational issues, such as determining order size, maintaining fill rate, and managing inventory levels (Tirkolaee *et al.*, 2022).

2.2.5.1 Information Technology

Information technology comprises both computers and digital communication tools that are capable of substantially decreasing the expenses related to communication and information procedures (Faheem et al., 2018). The alignment of information technology-related requirements with the business context is crucial, as it facilitates the enhancement of supply chain performance through the congruence between these requirements and the capabilities of information technology (Huo et al., 2015). In this investigation, in order to gain a more comprehensive understanding of the capital allocations made in the preliminary and subsequent stages, information technology has been classified into two distinct categories: one for suppliers and another for consumers.

Recent evidence has provided further validation of the notion that companies derive substantial advantages from the implementation of information technology. Profound productivity enhancements stemming from the seamless transmission and exchange of information have been uncovered (Baumann *et al.*, 2020). Other studies have presented compelling arguments suggesting that a sufficient amount of evidence has been accumulated regarding the beneficial outcomes of Information Technology (IT), thereby allowing for the classification of the productivity paradox as a historical fallacy (Mukhopadhyay *et al.*, 1997). There appears to exist a veracity that the investment in information systems does yield positive results at present, albeit the means by which it accomplishes this feat remains shrouded in enigma. Prior scholarly investigations have

acknowledged the negligence within IT value research regarding the interdependent impacts of IT in conjunction with other organizational facets, such as corporate strategies, customized production, and supply chain administration. Information software does not operate autonomously; it maintains a close collaborative association with other corporate assets (Andersen *et al.*, 2001). However, according to the studies conducted by Brooks and Davenport (2004) and Lou *et al.* (2004), it has been argued that technologies possess a considerable degree of uncertainty, which can be attributed to their fundamental characteristics such as autonomy (Jennings and Wooldridge, 1995), social ability (Moyaux and Chaib-draa, 2006), reactivity (Parunak, 1999), and pre-activeness (Moyaux and Chaib-draa, 2006). The assumption was made that the integration of information technology (IT) into the operations of the supply chain (SC) enhanced its ability to create value. IT possesses the capacity to oversee the movement of goods and services and to influence various aspects of the supply chain, including cost, quality, delivery, flexibility, and ultimately the financial gains of the organization (Baumann *et al.*, 2020).

Sanders *et al.* (2002) elucidated the explicit correlation between the utilization of technology in supply chain management. It was observed that organizations tend to utilize information technology more extensively than the standard practice in their respective industries, thereby attaining greater operational advantages, including cost reduction and cycle time optimization. Nevertheless, the efficacy of information technology employment in enhancing supplier network performance is contingent upon the industry's clock-speed (Guimaraes *et al.*, 2002). Narasimhan and Kim (2001) provided support for the notion that specific categories of information technology (IT) systems possess greater significance within the domain of supply chain management (SCM), as these systems aid organizations in enhancing their production and process control capabilities, efficiently managing pricing strategies, effectively addressing customer service concerns,

proficiently overseeing customer relations, optimizing inventory management, and streamlining warehouse operations. Material requirements planning (MRP) (C1) represents a production planning approach employed to facilitate the synchronization of material and resource availability with customer demand, thereby efficiently coordinating the fulfilment of orders (Koh, 2004).

The system could potentially yield more efficient resource allocation and decreased inventory levels by only issuing purchase and/or work orders when necessary. Enterprise resource planning (ERP) technologies have been specifically developed to tackle the issue of fragmented information within a business and promote integration with intra- and inter-enterprise data (Sharif *et al.*, 2005). Nonetheless, there exists a debate regarding the effectiveness of traditional ERP systems in facilitating real-time synchronization among partners within the supply chain, a crucial requirement for achieving efficient supply chain management (Karwowski *et al.*, 2007). The Advanced Planning System (APS) (C3) is designed to anticipate potential future consequences on the plan due to historical uncertainty patterns and potential future uncertainties. By utilizing this Information Systems (IS) practice effectively, it will lead to enhanced forecasting and more efficient resource planning, thereby improving operational efficiency (Lockhamy III and

McCormack, 2004).

Chae et al., (2005) demonstrated that the prevailing associations among the channel collaborators were subject to moderation due to the capacity of information technology to exert influence on inter-organizational collaboration. The capability of sharing information has a positive impact on the performance of the firm, particularly when there are elevated levels of both connectivity and willingness to share information (Fawcett et al., 2008). Thus, the implementation of information

technology (IT) to enhance connectivity among supply chain (SC) partners necessitates a dedication to information exchange in order to achieve improved performance. An effectively integrated IT infrastructure has the capability to present a comprehensive overview of the supply chain's current status, including the inventory levels of the manufacturers or their suppliers, as well as the service capabilities of the logistic providers. IT enables suppliers to access the inventory details of their customers and adequately prepare for timely stock delivery (Ngai *et al.*, 2010). It is contended, however, that the indeterminate effect of information technology (IT) on various facets of supply chain management (SCM) and unsatisfactory results of IT investment constitute a significant obstacle to the crucial contribution of IT in a company's performance (Ye and Farley, 2006; Kim *et al.*, 2006). With the aid of information technology, organizations have the capability to monitor market demands and efficiently reallocate resources in a timely and adaptive manner (Ngai *et al.*, 2010). Thus, the current investigation amalgamates the supplier relationship management system (SRM) with the customer relationship management systems (CRM). (Bayraktar *et al.*, 2009).

2.2.5.2 Relationship Management

Supplier Relationship Management (SRM) refers to the strategic planning and effective management of all engagements with suppliers with the aim of optimizing its inherent worth (Singh *et al.*, 2017). The organization additionally endeavours to identify novel providers with the aim of diminishing expenses, prognosticating procurement at a higher frequency, and exchanging information, all in order to attain various advantages through engaging in discourse with said suppliers. Cha and Kim (2018) stated that Supplier Relationship Management (SRM) is a field that involves cooperative collaboration with essential suppliers in order to enhance the potential value of the organization. SRM revolves around cultivating mutually advantageous

relationships with strategic supply partners, which yield higher levels of innovation and competitive edge compared to independent operations.

Klobučar and Erjavec (2019) posited that SRM managers ought to assume the responsibility of overseeing no more than three supplier relationships, so as to allocate ample time to each. Personnel engaged in SRM endeavours shall possess a commendable amalgamation of commercial, technical, and interpersonal proficiencies. Commercial expertise, knowledge of the market, the ability to analyse, and proficiency in project management are all significant. However, the essential aspect of developing robust and trustworthy working relationships lies in the mastery of "softer" skills such as effective communication, attentive listening, persuasive influence, and adept change management. Managers responsible for supplier relationship management possess a comprehensive understanding of their suppliers' business operations and strategic objectives, as well as the capability to perceive challenges from the supplier's perspective. Concurrently, they skilfully balance these considerations with the requirements and priorities of their own organization.

Kumar et al. (2021) reached a consensus that Supplier Relationship Management (SRM) comprises both business methodologies and software applications, constituting a fundamental element within the information flow aspect of supply chain management (SCM). The implementation of SRM practices aims to establish a shared foundation to facilitate efficient communication between an organization and its suppliers, even in scenarios where these parties employ disparate business approaches and terminologies. SRM, consequently, enhances the efficacy of procedures linked with the procurement of commodities and services, the administration of inventory, and the manipulation of resources. According to Liang et al. (2018),

the utilization of SRM software has the potential to result in decreased production expenditures and an improved level of excellence, although it may lead to a reduction in the price of the final product.

The process of customer relationship entails the organization obtaining information regarding their business from the customers, which is then utilized as a mechanism to effectively address their needs, while simultaneously striving to attain a profound comprehension of their requirements (Ziggers and Hensler, 2016). Customer Relationship Management (CRM) is described as the complete array of undertakings governed via the administration of customer engagement, the establishment of enduring commercial associations, and the augmentation of customer experience. This is perpetuated through the provision of client service and the establishment of connections with consumers (Chen and Wu, 2016; Quynh and Huy, 2018). The primary objective is to augment customer loyalty and affiliations in order to enhance the operational efficacy of the organization and furnish an exhaustive comprehension of the distinct requirements of consumers. Through the perpetuation of a robust customer relationship, an enterprise can distinguish itself from its contenders, safeguard consumer engagement, and amplify the value propositions extended to the clientele (Banerjee and Mishra, 2017).

CRM is considered to possess efficiency when it is integrated with the practice of SCM and the stage of decision-making (Yadollahinia *et al.*, 2018). The establishment of connections with consumers is imperative for the long-term viability of a company, particularly in light of the transition towards a phase characterized by heightened individualization and tailored service offerings (Banerjee and Mishra, 2017). The readiness of an organization to satisfy consumer demands and, as a result, adopt novel technologies or methodologies empowers the organization

to attain superior performance within a designated timeframe (Gandhi *et al.*, 2017). The notion that the client possesses an intimate alliance is being regarded as a fundamental aptitude of the enterprise and has the potential to manifest as a strategic advantage. In actuality, it is infrequent and unattainable for competitors to forge a personal rapport with clientele, thereby necessitating the enterprise and its stakeholders to attain heightened efficacy (Okongwu et al., 2015). CRM programs are typically implemented in organizations that possess an awareness of customer wants, needs, and brand devotion, while also evaluating the extent to which a product or service fulfills established criteria (Gawankar *et al.*, 2017). A business's capacity to accommodate consumer demands and ultimately innovate new products or processes empowers an organization to thrive in the long run (Hamister, 2012). In the presence of a customer relationship, the consumer will have access to the finest products and a wide selection of goods (Kaliani Sundram *et al.*, 2016). Thus, Customer Relationship Management (CRM) could potentially exert a substantial influence on the manner in which an organization manages its complete value chain (Gandhi *et al.*, 2017).

An efficient collaboration with suppliers and customers allows the entire supply chain to adapt to the dynamic market demands and respond appropriately. Simultaneously, personnel at different hierarchical levels within the producer and supplier organizations may offer diverse recommendations to continuously enhance and assess these relationships. Furthermore, they can obtain feedback and necessary information from customers and suppliers to facilitate decisionmaking. The findings of such investigations have showcased the imperative and indispensable nature of the association between individuals, collectives, and entities engaged in the supply chain in order to enhance its overall effectiveness and functionality (Alshurideh *et al.*, 2022).

2.2.5.3 Internal Integration

The term "integration" has been commonly referenced in previous research in regards to the notion of supply chain. More specifically, it has emerged as a novel perspective within the realm of business strategy linked to supply chain, aiming to establish a sustainable competitive advantage (Mani *et al.*, 2018). Other scholars have recognized that internal integration cohesion pertains to the degree to which an organization possesses the capability to transform its activities, trends, and behaviours into participatory, coordinated, and manageable processes in order to fulfil the demands of its consumers. This primarily encompasses cohesion in the information systems and data repositories. Furthermore, it establishes that diverse activities within an organization should not be approached individually, but rather as components of an integrated process (Khanuja and Jain, 2019).

The incorporation of information technology into corporate entities facilitates the achievement of internal unification, thereby yielding adaptability, transparency, accountability, and dependability (Tarigan *et al.*, 2021). Internal integration refers to the process of integrating the various departmental functions within an organization (Tarigan *et al.*, 2021). A company's internal assimilation with the implementation of information technology is employed for the consolidation of data, the manipulation of information, and the exploitation of resources (Gu *et al.*, 2021). Internal integration has the capability to retrieve data and information from other departments collectively and instantaneously (Lv *et al.*, 2018). The acquisition of knowledge from information technology systems can facilitate the process of making informed choices. The internal integration refers to the harmonization amongst procurement, production, manufacturing, finance, marketing, and other operational departments (Zhang *et al.*, 2018). A company's information technology

department will develop a strategic plan to address modifications in the upstream supply chain and the strategic responsiveness in the downstream supply chain, thereby transforming it into a collective capability (Lv et al., 2018). Internal integration is established within a single organization through the augmentation of the integration procedure. The process integration within the organization is executed through the implementation of collaborative planning and the cultivation of trust amongst the internal functions (Tarigan et al., 2021). Measurement indicators employed for evaluating internal integration comprise efficient data integration between different departments, expedited coordination across departments in response to alterations, prompt validation of data changes for other functions, timely integration of data during a pandemic, and timely availability of company data to all departments, among various other factors (Gu et al., 2021).

Internal integration refers to the incorporation of processes within various departments and functions as a manifestation of intra-organizational capabilities. In contrast to the conventional organizational structure characterized by fragmented and specialized departments, internal integration entails the dismantling of functional barriers and the cultivation of collaborative efforts and information exchange across all departments, with the ultimate objective of meeting customer demands. The exchange of information, the act of jointly formulating plans, regular communication and interaction, and the collective effort in reaching consensus are a collection of essential skills that enable organizations to cultivate a competitive edge. Furthermore, this collaborative endeavour also necessitates certain technological proficiencies in integrating information systems that are directly linked to operational effectiveness (Errassafi *et al.*, 2019).

In a cohesive supply chain, fostering a robust strategic alliance with suppliers will enable them to comprehend and foresee the needs of the manufacturer, thereby enhancing their ability to effectively fulfil its evolving demands (Flynn *et al.*, 2010). As an example of inter-organizational capabilities, the act of sharing information pertaining to products, processes, production schedules, and plans, in addition to collaborating with suppliers, aids in the synchronization of their respective production plans and the reduction of lead time. Through the establishment of a common comprehension of the manufacturer's operations and processes, suppliers attain a heightened level of customer service, subsequently assisting manufacturers in enhancing their own customer service (Flynn *et al.*, 2010).

2.2.4 Sustainable Supply Chain Management

The notion of Supply Chain and Sustainability Management (SSCM) has gained considerable attention among professionals and scholars alike. This is primarily attributed to a multitude of factors that contribute to its embrace and endorsement. These factors include the demands articulated by stakeholders, the elevated expectations of customers, the imposition of legislative measures by governments, the societal expectations regarding the corporate image and reputation, the competitive pressures exerted by rivals, the concerns pertaining to environmental pollution and the depletion of natural resources, among other considerations.

The term SSCM encompasses the incorporation of sustainability perspectives into the definition of supply chain management (SCM) and is extensively discussed in the literature under various labels such as green supply chain management (GSCM), ethical or responsible SCM, and corporate social responsibility (CSR), among others (Broman *et al.*, 2017; Köksal *et al.*, 2017). Gupta *et al.* (2019) provided a definition for SSCM which encompasses the strategic and

transparent alignment of an organization's social, environmental, and economic objectives. This alignment is achieved through the systematic coordination of crucial inter-organizational business processes, ultimately leading to the enhancement of the individual organization's long-term economic performance as well as that of its supply chains. The authors emphasize the necessity of inter-organizational coordination, with a primary focus on achieving sustainable long-term economic outcomes. According to Herrera *et al.* (2020), a genuinely sustainable supply chain is one that does not impart any negative impact upon the environment or society. Moreover, it is imperative that the non-economic aspects, such as the environmental and social dimensions, are in harmony with the economic dimension, and vice versa.

Organizations must undertake the task of reconfiguring their existing supply chains in order to integrate sustainability objectives throughout their operations, ranging from procurement to distribution, in order to attain sustainability (Allaoui *et al.*, 2019). Ben Abdelaziz *et al.* (2015) had determined the objectives of a Sustainable Supply Chain Management (SSCM), which include the optimization of value for all parties involved and the satisfaction of customer demands through the establishment of sustainable streams of goods, services, information, and financial resources. Moreover, it aims to facilitate collaboration among the various participants within the supply chain.

SSCM is widely regarded as the strategic and harmonized coordination and achievement of an organization's societal, environmental, and economic objectives through the systematic administration of fundamental inter-organizational market procedures, with the aim of augmenting the enduring economic prosperity of the specific corporation and its supply chains (Busse *et al.*, 2017). The process incorporates measures that are put into practice to enhance the

durability of the system over an extended period of time, encompassing the interconnectedness between the various elements and engagements within the network of product distribution. The shift from Supply Chain Management (SCM) to Sustainable Supply Chain Management (SSCM) places significant pressure on organizations to modify their existing supply chains in order to meet the requirements for sustainability (Schrettle *et al.*, 2014).

Organizations contribute to the advancement and enhancement of sustainable supply chain management (SSCM) through the provision of enduring advantages and a competitive edge for economically and socially advantageous endeayours throughout the entirety of the supply chain (Ahi and Searcy, 2013; Zhu and Sarkis, 2006). Control of the supply chain, however, entails a constantly dynamic task. (Ageron et al., 2012). The execution of Supply Chain and Sustainable Management (SSCM) initiatives promotes enhanced substance and rigorous upkeep (Gunasekaran and Spalanzani, 2012), as well as originality, amplifies the financial fruition of enterprises (Wang and Sarkis, 2013), and establishes a brand reputation grounded in market-based principles (Zailani et al., 2012). When addressing matters pertaining to environmental issues, enterprises and scholars employ the utilization of green supply chain management (SCM) or environmental sustainability frameworks with the objective of mitigating detrimental environmental impacts (Ben Brik et al., 2013; Mathiyazhagan et al., 2015), like as effectiveness of resource use, recycling and disposal, waste and carbon mitigation (Koberg and Longoni, 2019). The integration of environmental concerns into supply chain management has been identified as contingent upon effective collaboration and coordination among the various stakeholders involved (Zhu et al., 2010).

2.2.7 Top Management

Top-level executive dedication in the management of supply chains has emerged as a pivotal factor due to its criticality in ensuring the triumphant execution of supply chain management. The highest echelons of management foster the facilitation of employee empowerment and heightened job contentment by means of astute guidance, ultimately aiming to accomplish the objective of customer contentment (Sajjad *et al.*, 2020). The upper echelons of leadership consistently endeavour to execute the Supply Chain Management (SCM) system within the organization through assuming accountability and jurisdiction over the implementation of an efficient SCM system. The company's top management guarantees that the formulation of policies and the establishment of objectives are tailored to both the internal and external circumstances of the company (Tien *et al.*, 2019).

The top management of the organization ensures the establishment of a harmonious relationship between the internal and external stakeholders. This integration is designed to facilitate a streamlined and highly productive flow of supply chain management activities. Additionally, the management of the company guarantees the availability of the requisite resources to effectively implement supply chain management in accordance with the specified criteria. The management engages in effective communication and fosters coordination and collaboration with the stakeholders participating in the establishment of the supply chain management system within the organization. The management must possess the ability to control the internal circumstances of the company to facilitate the involvement and empowerment of all elements within the organization, thereby contributing to the enhancement of company performance through the practical implementation of supply chain management principles (Fischer *et al.*, 2020; Prajogo, *et al.*, 2018).

2.2.8 Challenges Affecting Efficient Supply Chain Management

The obstacles encountered by Supply Chain Management (SCM) in both its theoretical and practical aspects arise from the complex interaction and lack of alignment between them (Golgeci and Gligor, 2017). One primary obstacle lies in questioning the notion of "managing" the supply chain. Another obstacle is to expand the range of involvement in supply chain management (SCM) - the "arc of integration". This can only be accomplished through more efficient utilization of the identified facilitators – the enhanced clarity of information and knowledge, the establishment of suitable relationships, and the creation and utilization of suitable metrics (Ali et al., 2021). The logistics function is merely one facet among numerous domains that exert influence on the efficacy of the supply chain within the cocoa industry. Integrated process management, information systems and the dissemination of information, reorganization of the organization, and cultural realignment are all equally significant. The necessity for seamless integration of processes spanning from the procurement of raw materials to the ultimate delivery of the finished product is vital to the triumph of a company. Merely possessing a high level of manufacturing efficiency no longer guarantees a competitive advantage (Amankwah-Amoah et al., 2018). The utilization of integrated planning throughout the supply chain is not adequately implemented by the industry. The presence of this form of disunity within the supply chain may result in a rise in the expenses associated with the procurement of crude oil, consequently impacting the prices of gasoline for consumers (Bai et al., 2022).

Due to the pervasive global reach of the petroleum industry's supply chain, the implementation of advanced information technology is of utmost importance in ensuring the seamless flow of information, given the intricate nature of the logistics network within this sector. The effectiveness

of companies' engagements within supply chain networks is intrinsically tied to the proficient utilization of information technology (Piya *et al.*, 2017). The enhancement of operational effectiveness within the petroleum sector necessitates the adoption of a novel conceptual framework for collaboration, even if this entails cooperating with industry rivals. The crucial alteration in mind-set must be directed towards fostering collaboration, exchanging information, and optimizing assets, as chemical manufacturers and Logistics Service Providers (LSPs) would be required to collaborate with both their competitors and other entities within the supply chain. While the acquisition of advanced information technology is indispensable, its effectiveness will be limited without the accompanying cultural transformation (Schultz *et al.*, 2021).

2.3 Theoretical Review

SCM practices hold significant importance for any organization due to their ability to enhance an organization's market leadership, profitability, and overall strategic positioning by effectively manipulating market variables such as price, cost, quality, delivery, and product innovation, among other factors (Arshad Ali *et al.*, 2020). Three theories provide an explanation for the concept of Supply Chain Management (SCM) practices: Systems theory, Contingency theory, and Resource-Based View Theory.

2.3.1 Systems Theory

Scholars specializing in systems theory assert that all issues in the fields of natural and social sciences possess an inherent systemic essence (Hutchins, 1996; Meadows, 2008; Wheatley, 2006). The application of systems theory involves a comprehensive explanation of problems, contrasting

with the conventional approach of Western science. Banathy (2006) contended that due to the phenomenon of specialization in science that emerged in the 17th century Scientific Revolution, numerous researchers have become confined within their own distinct domains (2006). Ever since Descartes, scientific dilemmas have been resolved by dissecting them into smaller components and continuously reducing their complexity. The crux of conventional scientific investigation lies in the utilization of the scientific method and the application of

controlled settings to isolate and manipulate variables.

This particular theory posits that the perception of an occurrence is perceived in its entirety, rather than being contingent upon its constituent subsystems (Choi *et al.*, 2018). A system consists of subsystems that interact and rely on each other to achieve a state of balance within a larger system (Amagoh, 2016). The primary emphasis is placed on the interconnectedness of subsystems, enabling a more comprehensive comprehension of an entity's structure, operations, and outcomes. Additionally, this perspective recognizes that an organization's functioning is influenced by the external environment, involving a range of entities such as agents, shareholders, and other factors that lie beyond the organization's control (Sanderson *et al.*, 2015). Systems theory integrates multiple supply chain factors, which subsequently shape a more extensive network of supply chain systems. Furthermore, it aids in uncovering the degree of interdependence among the elements of the system, thus enhancing comprehension of the dynamics of the supply chain. Consequently, the planning, execution, and coordination of manufacturing company activities are improved (Jaradat *et al.*, 2017).

WU SANE NO

2.3.2 Contingency Theory

Contingency theory is a scholarly perspective employed to examine the behaviour of organizations. Within this framework, elucidations are provided regarding the impact of contingent factors, such as technology, culture, and the external environment, on the configuration and operation of organizations. The fundamental assumption that underlies contingency theory posits that no singular organizational structure possesses equal applicability to all organizations. In contrast, organizational effectiveness is contingent upon the congruity or concordance amidst the technology type, environmental volatility, organizational size, organizational structure characteristics, and its information system. The theories pertaining to contingency have been originated from the sociological functionalist theories of organizational structure, namely, the structural approaches to organizational studies as proposed by Reid and Smith (2000), Chenhall (2003), and Woods (2009). The aforementioned studies posited the conjecture that the establishment of an organizational structure was contingent upon contextual factors, including but not limited to technological advancements, dimensions of the task environment, and the size of the organization. In specific other scholarly works, the theory of contingency maintained its status as a prominent paradigm in the realm of management accounting research (Cadez and Guilding, 2008).

The theory posits that in various circumstances, disparate resolutions may demonstrate significance (Yu et al., 2020). Instead of adhering to conventional management principles, the theory aims to illustrate that varying circumstances necessitate distinct organizational frameworks and infrastructure. Organizations encounter constraints stemming from various factors, such as the scale of the enterprise, the surrounding environment, and the utilization of information

technology. These contingencies are intended to cultivate the particular structures and operations of an organization (Morais and Barbieri, 2022).

Hanelt *et al.* (2021) postulated that in order to enhance the operational capability to generate revolutionary commodities, a corporation should modify its organizational attributes and structure its pivotal constituents to establish a resilient and adaptable supply chain (Bag *et al.*, 2021).

2.3.3 Resource Based View Theory

The resource-based viewpoint (RBV) posits that a company's enduring competitive edge is contingent upon its invaluable, uncommon, unreplaceable, and irreplaceable assets (Barney, 1991). The capacity of companies to generate or procure these assets influences their performance and competitiveness vis-à-vis their rivals. The resource-based perspective advocates for the importance of resources that are unique to a specific company, specifically those resources that maintain value within the company's markets and other resources that are difficult to replicate by other companies (Wernerfelt, 1984). These resources include managerial skill, customer relationships, brand reputation, and tacit knowledge related to specific manufacturing processes. Resources can be differentiated from competencies or capabilities. Conversely, a firm's ability to obtain resources and its capability to assemble and combine these resources in particular ways determine the firm's expertise in a specific product domain. When a firm acquires resources for a certain business, these resources may, to differing extents, be sufficiently adaptable for use in other product lines or markets (Teece, 1982).

This theory emphasizes the benefits that a company obtains through the possession of essential resources necessary for its sustenance. These resources can manifest in the form of financial

SANE

strength, physical facilities, human labour and dedication, technological progress, and other capacities. These resources and capacities distinguish a firm from others and constitute a fundamental aspect of its competitive advantage. The possession of products and services with distinct attributes or specific and detailed work procedures will exclude competition for the resources and capacities of the firm (Antoni and Akbar, 2019).

The ability of organizations to adjust to changes in the industrial and market landscape can also be viewed as a favorable circumstance to address the challenges posed by global competition. A well-managed supply chain plays a crucial role in establishing a competitive edge and generating value (Madhani, 2019). The nature of competition has evolved from being limited to individual companies to encompassing the competition between supply chains (Min *et al.*, 2019). The advantages of this theory, consequently, are directly gained by the enhancement of operational performance (Varadarajan, 2020).

2.4 Empirical Review

Hong *et al.* (2018) with the application of a Structural Equation modelling (SEM) analysed data from 209 manufacturing firms in China with the aim to investigate the effect of Sustainable Supply Chain Management (SSCM) practices on the dynamic capabilities of the supply chain and as well on enterprise performance. Results revealed a significant positive effect of SSCM on the dynamic capabilities of supply chain and the dimensions of performance. Results also revealed that the relationship between SSCM practices and enterprise performance was partially mediated by SC dynamic capabilities.

With a focus on sustainable supply chain and supply chain risk, Gouda and Saranga (2018) analysed data from six selected manufacturing sectors in 21 different countries. Through the

application of a Partial Least Squares Structural Equation Modelling (PLS-SEM), results revealed that risk mitigation strategies employed did not always result in the reduction of supply chain risk more specifically in emerging markets. Results further revealed that risk mitigation strategies were rather effective when utilised in conjunction with sustainable efforts although they fail to reduce supply chain risk on their own.

Isnaini et al. (2020) conducted a study to determine the relationship that exists between Sustainable Supply Chain Management and Sustainable Organisational Performance employing supply chain dynamic capabilities as a moderating factor. The study utilised a PLS Sequential Equation Modelling (SEM) method to analyse data obtained from 210 supply chain managers in Indonesia. Results revealed that there was a significant relationship between SSCM practices and the indicators of sustainable organisational performance. Results further indicated that there was a significant moderation of supply chain dynamic capabilities on the relationship between SSCM practices and the indicators of sustainable organisational performance.

Golicic and Smith (2013) through the application of a meta-analysis sought to determine the impact of Sustainable Supply Chain Management practices on firm performance. Through the review of research spanning 20 years, the findings have unveiled that there exists a constructive and noteworthy correlation between the implementation of environmental supply chain practices and the various forms of firm performance, namely market-based, operational-based, and accounting-based. With the application of firm size and time as moderating variables, results revealed significant influence on the relationship between sustainable supply chain management and firm performance environmentally.

Reinartz *et al.* (2004) aimed to ascertain whether the integration of SRM procedures is positively correlated with organizational effectiveness. The results suggest that the incorporation of SRM procedures is connected with enhanced company performance in two phases. The most impactful outcome is observed in relationship preservation, followed by relationship initiation. The influence of relationship termination was either minimal or not statistically significant. Hence, it can be inferred that SRM yields some of the anticipated benefits that companies anticipate upon investing in it.

Shin et al. (2000) made an endeavour to scrutinize the impact of supply management on the operational efficacy of suppliers and the competitive priorities of buyers, specifically encompassing cost, quality, delivery, and flexibility. The findings of the investigation substantiate the assertion that an enhancement in supplier relationship management yields positive outcomes for both suppliers and buyers, thereby establishing a mutually beneficial scenario within the supply chain. Moreover, the impact of SRM on performance that is associated with delivery and quality surpasses its impact on cost or flexibility performance in terms of statistical significance. Indeed, when the competitive priorities of volume and process flexibility are of utmost importance, adopting a supply chain management approach may not yield the desired level of flexibility.

TENS AND SANE

2.5 Conceptual Framework

Conceptual framework is the system of connected ideas that together gives a far reaching comprehension of an idea (Durso *et al.*, 2017). Algozzine and Hancock (2017) delineate a conceptual framework as a concise portrayal of the concept being studied, accompanied by a graphical or visual representation of the key elements of the investigation. It is an outline that demonstrates the connection between your independent factors and the dependent factors.

Figure 2.1: Conceptual Framework Moderating Variable Top Management Commitment Independent Variables **Efficient SCM** H H Information Dependent Technology Variable H Sustainable Supply Relationship 2 Chain Management H Internal Integration

The concept is divided into two sections, the efficient supply chain management practices which is the independent variable comprising of key practices such as information technology, customer

Source: Adopted from Daneshvar et al. (2020).

relation management which is divided into customer and supplier relationships, and internal integration. The dependent variable constitutes the sustainable supply chain.

In the sector dealing in a business of cocoa, efficient supply chain management practices can be a solution to many factors that might be facing such organizations including supplier and customer relationship management. Management will therefore need to examine some of the practices, strategies and assess the risk associated with the implementation of such policy. A comprehensive approach towards the implementation of information technology, relationship management as well as internal integration practices relative to supply chain sustainability can help answer questions pertaining the productivity or performance of the companies or organisations. Management of the organisations can choose a type of efficient supply chain management strategy they deem fit for the company. Information technology, relationship management, and internal integration practices aligns with questions seeking the key efficient supply chain strategies to be implemented.

The conceptual framework thus endeavours to elucidate the researchers' integration of the literature delineated in light of the research objectives and the research query.

2.5.1 Relationship Between Information Technology and Sustainable Supply Chain

ICT is widely recognized as a facilitator of increasingly sustainable supply chains (Oh *et al.*, 2019). Moreover, it serves as a catalyst that can potentially enhance sustainability through the stimulation of groundbreaking innovations (Melander and Pazirandeh, 2019). The achievement of economic sustainability is frequently attributed to the process of digitalization, which creates opportunities for the execution of projects centered around socio-environmental technology.

While the utilization of information and communication technology (ICT) and digitalization to enhance sustainability is still in its early stages (Thöni and Tjoa, 2017), the potential provided by ICT to advance sustainability in supply chains has been acknowledged (Govindan *et al.*, 2018). Collaboration among supply chain participants through the utilization of ICT applications might potentially play a crucial role in attaining sustainability objectives (Bag *et al.*, 2018). Additionally, the amalgamation of information and the exchange of data among organizations may facilitate the implementation of more environmentally-friendly cooperative methods within the logistics network (Beier *et al.*, 2018). Therefore, it is hypothesised that:

H₁: Information Technology have a positive significant impact on sustainable supply chain.

2.5.2 Relationship Between Relationship Management and Sustainable Supply Chain

Various research investigations have examined the ways in which sustainable methodologies can bolster connections with clientele. Chen et al. (2020) demonstrated through their investigation the influence of presumed sustainability measures on the level of consumer involvement in online shopping. Gil-Gomez et al. (2020) have acknowledged the potential of fostering consumer loyalty by making substantial investments in the customer relationship, which in turn contributes to the long-term viability of organizations. Mena et al. (2019) have demonstrated that companies employing a market-oriented strategy have a propensity to meet consumer needs through the provision of appropriate products and services at reasonable prices, resulting in enhanced business relationships and expeditious outcomes. Challenges associated with the sustainability of client relationships encompass the ability to cultivate customer loyalty by means of heightened allegiance, facilitated by the provision of amiable goods and services, as well as the positive regard and prestige associated with catering to environmentally conscious consumer behaviour (Wikhamn, 2019). Sundram et al. (2011) asserted that within the realm of supply chain

management, when consumer engagement is operationalized, the customer will receive products and services that meet their expectations in terms of quality and variety. Thus, hypothesis H₂ is stated as follows:

H₂: Relationship management have a positive significant impact on sustainable supply chain.

2.5.3 Relationship Between Internal Integration and Sustainable Supply Chain

Companies established internal integration to facilitate collaboration with both suppliers and customers. The integration within 539 Taiwanese third-party logistics (3PLs) has a positive effect on the integration with customers and the collaboration with logistics partners (Liu and Lee, 2018). The utilization of companies' information technology for the purpose of internal integration has a significant effect on the partnerships within the supply chain, particularly in terms of the flow of raw materials and the sharing of information (Gružauskas and Vilkas, 2017). The formation of internal integration within a company allows for the connection between the company itself and its suppliers, thereby fostering SC partnership (Tarigan *et al.*, 2020). The coordination and integration between manufacturing companies and suppliers to establish a purchasing strategy can be affected by the internal integration within these companies. This internal integration is achieved through the implementation of enterprise resource planning (ERP) as a unified database system (Tarigan *et al.*, 2020).

The capacity of the organization to adapt internal assets in order to swiftly address alterations significantly influences the enhancement of supply chain sustainability (Ambulkar *et al.*, 2015). Internal collaboration among the various departments within the organization, such as procurement, production, sales, and finance, consistently collaborates to establish corporate objectives with the aim of enhancing the sustainability of the supply chain (Piprani *et al.*, 2020).

Internal integration, as suggested in the collection of data, the processing of information, and its utilization, is connected to the sustainability of the supply chain (Gružauskas and Vilkas, 2017). The quality of integration achieved within the supply chain for logistics services can potentially have a positive influence on the sustainability of the supply chain (Ju *et al.*, 2020). Sharing of information is a crucial element that businesses must prioritize in order to enhance the sustainability of their supply chains (Hohenstein *et al.*, 2015). Sharing information among company members, as a means of supporting strategy for enhanced management, has the potential to augment the sustainability of the supply chain (Karmaker and Ahmed, 2020; Hosseini *et al.*, 2019). Thus, it is hypothesised that:

H3: Internal Integration have a positive significant impact on sustainable supply chain.

2.5.4 Moderating Effect of Top Management Commitment

The presence of top-level executives plays a moderating role in the relationship between supply chain performance and the integration of internal processes, external processes, and internal processes with suppliers. The creation of social value contributes positively to the preservation of the environment (Li *et al.*, 2022). The community is expected to thrive under the China-Pakistan Economic Corridor (Aman *et al.*, 2022; Hussain *et al.*, 2019; Abbas *et al.*, 2020). Furthermore, it is imperative that individuals are dissuaded from engaging in tax evasion in order to enhance the viability of enterprises (Li *et al.*, 2022). It is imperative to acknowledge, participate in, and proficiently oversee these issues in order to maximize the vast capacity for accomplishment that is inherent in the amalgamation of numerous diverse aspects of sustainability. Following the adjustment for variables such as the scale of the organization, viewpoint on governmental regulations, financial constraints imposed by the organization, and prevailing practices within the industry at different tiers, Suryanto *et al.* (2018) ascertained that mutual associations of

organizational learning mechanisms, organizational support, and adoption of GSCM practices were found to be both positive and substantial. Thorough investigations have been undertaken to examine the repercussions and ramifications of ineffective logistics management. Sajjad *et al.* (2020) conducted an examination that showcased the instrumental and normative justifications for the utilization of SSCM. Thus, it is hypothesised that:

H4: Top management commitment moderates the relationship between efficient supply chain management and sustainable supply chain.

2.6 Summary

The literature review examined various domains that the researcher considered highly appropriate to the study's objectives. The concepts reviewed focused on supply chain, supply chain management, and sustainable supply chain management practices. The theories relevant to the study reviewed included the systems theory, contingency theory, and resource based view theory.

The Conceptual framework was developed with the elements of information technology, relationship management, and internal integration as independent variables and sustainable supply chain as the dependent variable. These discussions have contributed to the explanation of efficient practices in the management of the supply chain. This is due to the fact that supply chain management is a multifaceted concept, and there is no singular theory or set of theories that can fully capture and explain this concept.

CHAPTER THREE RESEARCH METHODOLOGY

3.1 Introduction

This particular segment of the investigation encompasses the methodologies and procedures that were utilized in the acquisition of the data. The research methodology delineates the overall approach of a meticulously crafted procedure employed for the compilation of dependable and precise information, with the intention of conducting an inquiry. It is a systematic series of gathering information from a cohort of individuals in order to gain insight into a specific circumstance or event within that cohort. It encompasses the research blueprint, the demographic of the study, the sample size, the research methodology and plan, the data sources and sampling methodology, the process for data collection and analysis, the ethical considerations, and the reliability of the instruments.

3.2 Research Design

Bryman (2017) outlines five distinct categories of research methodologies, encompassing experimental design, cross-sectional or survey design, longitudinal design, case study design, and comparative design. This research study is basically focused on a case study approach utilising staff members of manufacturing companies within the Ashanti Region of Ghana. The primary aim of employing a case study methodology is typically to generate a comprehensive examination of a particular case, followed by conducting a theoretical exploration of the subject matter (Bryman, 2017).

According to Creswell and Creswell (2017), a case study could be described as a type of research strategy adopted to investigate and ascertain a detailed analysis of a situation. This strategy enables the research study to focus on a subject matter that is of interest, which in this case is

students' perception, and then also affords the researcher the opportunity to examine more closely and into detail a specific group. According to Van Thiel (2014), this type of research strategy focuses on examining situations happening every day and in real life. In the views of Rashid *et al.* (2019), a case study is a systematic study conducted in small groups so as to present a better understanding and appreciation of a bigger group with comparable characteristics. This is to say that the case study approach concentrates solely on the situation as hand, in order to derive a comprehensive understanding from a distinctive circumstance or specific problem. Thus, the case study strategy is adopted for the purposes of this study to help examine the impact of efficient SCM practices on sustainable supply chain within the cocoa sector of Ghana.

3.3 Study Population

The population for a study is the total collection of elements on which some inferences is to be made about (Becker, 2017). In essence, the population is the large collection of subjects or elements from which a representative sample is drawn from (Etikan *et al.*, 2016). The targeted population for the study is all cocoa companies listed on the Ghana Stock Exchange (GSE) as at December, 2022. Notwithstanding, the study needed to reduce the population to a workable and manageable size due to limitations on time and finances with the use of appropriate sampling techniques discussed below.

3.4 Sampling Technique and Sampling Size

The necessity to engage in the process of sampling is one that is nearly always encountered in the realm of scholarly inquiry (Bryman, 2017). The size of the sample has been defined as the subset of the overall population that is deliberately chosen to serve as a representative sample (Neuman, 2014). The primary objective of sampling in research is to generate a representative sample that

accurately reproduces or reflects the pertinent characteristics found within a larger assemblage of instances known as the population (Creswell and Creswell, 2017). In the context of this particular investigation, the act of sampling entails the deliberate selection of certain elements within the target population, with the intention of utilizing the data gathered from these elements to gain a comprehensive understanding of the entire population.

First, a convenience sampling was employed in the selection of 5 cocoa companies within the Kumasi metropolis of the Ashanti Region that are in close proximity of the researcher and then, the researchers utilized the simple random sampling technique to determine the 200 staff members from the chosen companies. They believed that these individuals possess the necessary experience for the study and were able to allocate enough time and willingness to participate (Taherdoost, 2017). This technique, known as simple random sampling, entails gathering responses from willing participants within the sample frame who are also available for the study.

The simple random sampling technique which is one of the main types of probability sampling methods made up of people who are easy to reach, was also employed in the selection of respondents during the survey process, by this method; Senior staff members who were readily available were selected for answering of the questionnaires. The advantage this technique presents is that, it affords the respondents the luxury to decide on their own volition to participate in the survey and therefore will not be forcefully selected against their wish. This technique was also selected because of its ability to boost the participation or response rate of since workers in general are sceptical in participating in surveys they thick could have impact on their employment status.

3.5 Method of Data Collection

The instruments used by the researcher in collecting primary data from the respondents for analysis was a self – administrated questionnaire with closed-ended questions. All respondents were given the same set of self-administered questionnaires. Bernard (2017) posits that the purpose of a questionnaire is to target a particular objective, research question, or hypothesis. In this particular study, a questionnaire was employed due to its capacity to swiftly gather a significant volume of information. The questionnaire was meticulously developed and organized to encompass all the variables relevant to the study, in order to fulfil the study's objectives.

3.6 Method of Data Analysis

Silverman (2018) provided a definition for data analysis, which involves the systematic organization, arrangement, and interpretation of a large volume of gathered information. The data gathered for the investigation possessed a quantitative nature. The examination of the data collected for this inquiry involved the utilization of descriptive and inferential statistics. According to the study conducted by Ho and Yu (2015), the application of descriptive statistics permits the researcher to depict the distribution of scores. Frost (2018) affirmed that inferential statistics is a method that identifies the variable that either predicts or offers the most comprehensive explanation for the portion of the total variance found within the scores of dependent constructs. By choosing this approach, the researcher was able to effectively portray the distribution of scores by employing statistical measures that are contingent upon the types of variables examined and the scale of measurement.

The data was gathered through the utilization of a self-administered survey. The data was subsequently arranged and encoded, with descriptive statistics encompassing frequencies and measures of central tendency, such as the mean and standard deviation, being employed to present

the data using the Statistical Package for Social Sciences (SPSS Version, 22). Additionally, inferential statistics were executed to ascertain the degree to which the research variables influence the dependent variable.

3.7 Reliability and Validity

Two fundamental principles in the measurement of social research quality are the concepts of validity and reliability (Mohajan, 2017). The qualitative research field exhibits greater flexibility in regards to reliability and validity when compared to quantitative research. The necessity of ensuring reliability and validity in the study arises from its objective to examine the significance of numerous informal relationships, while considering relevant information derived from individuals' perspectives, which is regarded as a means to establish the validity of the study. Moreover, this study employs the collection of information from multiple viewpoints as the central approach to attaining reliability and validity, thereby enabling comparisons to be made. A potential challenge regarding validity may arise due to the possibility that respondents may exhibit bias when presenting their perspectives. In order to uphold the reliability and validity of the study, the information obtained from transcriptions was provided to respondents for the purpose of verification and confirmation. Additionally, the researcher utilized Cronbach's Alpha in order to assess the reliability of the proposed constructs. Renowned for its stability and flexibility, Cronbach's alpha serves as a measure of internal consistency or interconnectedness among the items (Vaske *et al.*, 2017).

3.8 Ethical Consideration

It is necessary that considerations are given to issues of ethical standards especially in the design of a case study research approach since it mostly involve issues of confidentiality about an individual or an institution. According to Fleming and Zegwaard (2018), issues of ethical

considerations comprise of receiving informed consent, receiving approval from a review board, managing sensitive information, provision of stimuli and feedback retrieval. In line with this, ethics procedures, guidelines and conduct in relation to confidentially, anonymity, voluntary participation and informed consent were adhered to. The researcher first sorts the permission from some of the Management of the selected companies with an introduction letter to undertake a study within working hours.

Kadam (2017) explicated the notion of informed consent as a collaborative process that transpires between the researcher and the participants of the sample, rather than being a solitary occurrence. Prior to engaging in the research, potential participants were thoroughly apprised of the procedures and potential risks associated with the study, after which they willingly provided their consent to partake. Over the course of the investigation, participants were consistently apprised of the research inquiries, purpose, and objectives, and were subsequently invited to participate on a voluntary basis and they were also made to be aware that the research was for academic purposes. To ensure confidentiality and anonymity, respondent's identity was cancelled throughout so that responses could not be linked to respondent in the case study organization.

CHAPTER FOUR DATA ANALYSIS AND PRESENTATION

4.1 Introduction

This chapter provides an exposition and depiction of the examination and findings of the data with regards to the objectives of the research. The scrutiny was conducted on the basis of the components in the survey in order to facilitate the study in arriving at substantial deductions while concurrently assisting in addressing the inquiries of the study. This segment is classified into four principal divisions, specifically; the demographic traits of the participants, the practices of supply

chain management, the capacity of resources, the performance of the supply chain, and the hurdles faced in supply chain management. In total, 200 questionnaires were administered but only 167 were retrieved resulting in a response rate of 83.5%.

4.2 Demographic Characteristics of Respondents

This section pertains to the fundamental data information of the individuals surveyed. In the pursuit of attaining a comprehensive comprehension of the inquiries emanating from the issue, it was imperative to obtain a bio-data of the respondents to effectively contextualize their diverse concerns. This includes age, gender, length of service, level of education, level of company experience, designation, and staff strength of organisations which are presented as follows;

Table 4.1 above reveals the age of respondents. Majority of the respondents fall within the age category of 31-40 years representing 52.1 percent signifying 87 of the total respondents. This is followed by 20-30 years representing 27.5% (46). The survey however recorded fewer participant within the age range of 41-50 years representing 12.6% (21), and above 50 years representing 7.8% (13). Results is in agreement with information from the Ghana Statistical Service which reveals that majority of the Ghanaian population who are engaged within the supply chain industry of Ghana are the youth between the ages of 20 and 40 years (GSS, 2021).

Table 4.1: Age of Respondents

| Response | Frequency | Percentage (%) |
|----------|-----------|----------------|
| 20-30 | 46 | 27.5 |
| 31-40 | 87 | 52.1 |
| 41-50 | 21 | 12.6 |
| Above 50 | 13 | 7.8 |
| Total | 167 | 100 |

Source: Fieldwork, 2023

Table 4.2: Gender of Respondents

ASAP2 A

| Response | Frequency | Percentage (%) |
|----------|-----------|----------------|
| Male | 106 | 63.5 |
| Female | 61 | 36.5 |
| Total | 167 | 100 |

Source: Fieldwork, 2023

Men and women possess a propensity to interpret matters through contrasting viewpoints, hence necessitating the identification of the gender of the participants. The male population, consisting of 106 individuals, constitutes the majority with a percentage of 63.5, whereas the female population, comprising 61 individuals, constitutes 36.5 percent. Explanation can be given to the large numbers of male respondents as a result of the companies being a manufacturing related ones hence majority of the 36.5 percent female can be deduced to be part of the respondents belonging to the management and supervisory roles of the company. This was in similarity to data provided by GSS (2021) that 0.2 percent of females as against 0.6 percent males are employed in the supply chain sector. The results again give the suggestion that fewer females are in management roles. The gender distribution of the respondents is presented in table 4.2 above.

Professional certificate

H. N. D.

Undergraduate

Postgraduate

No formal education

0 10 20 30 40 50 60 70

Figure 4.2: Education Level of Respondents

Source: Fieldwork, 2023

Figure 4.1 above gives indications about the educational levels of the respondents revealing that majority of the respondents (67) representing 39.9 percent of the total respondents have

undergraduate qualification as at the time of the survey. 51 and 34 respondents representing 30.5 percent and 20.6 percent respectively have a Higher National Diploma (H. N. D.) and postgraduate qualification. 15 respondents representing approximately 9% of the total surveyed respondents had a professional certificate. No respondent was recorded to have had no form of a formal education. These results affirm the assertions made by Hee and Jing (2018) stating that majority of the staff of the Ghanaian population working in the manufacturing sector have a tertiary level qualification.

Table 4.3: Length of Service

| Response | Frequency | Percentage (%) |
|----------|-----------|----------------|

| Over 20 years Total | 24 167 | 14.3 100 |
|-------------------------------|-----------|--------------|
| · | | |
| 5 - 10 years 11 - 15 years | 37 34 | 22.4 20.2 |
| Under 5 years | 55 | 32.7 |

The table above (table 4.3) presents the length of service distribution of the respondents. The table reveals that majority of the respondents have been in service under 5 years representing 32.7% (55) of the total number of respondents. The second highest category of respondents belong to the group of between 5 to 10 years representing 22.4% (37) of the total respondents. The category of workers that falls within 11-15 years is represented by 20.2% (34) of the total number of respondents. 16-20 years' category is represented by 10.3% (17) of the total number of respondents. Workers who had over 20 years' length of service with their respective companies numbered 24 respondents representing 14.3%. This is in agreement with data from the Ghana Living Standard Survey which gives the indication that majority of the working population in Ghana are the youth and the young people between the age range of 15 and 35 representing a cumulative percentage of 61.9 (GLSS, 2021).

THE WAS ANE

Respondents' Designation Senior Manager Manager **]** 24 Supervisor 52 Senior Staff 61 Junior Staff 10 20 30 40 50 60 70

Figure 4.3: Designation of Respondents

Source: Fieldwork, 2023

In terms of respondents' distributions relative to their designation, Figure 4.2 reveals that majority of the respondents surveyed numbering 61 were junior staff. This is followed by senior staff who numbered 52. Supervisors who were surveyed numbered 24, whereas managers and senior managers who were surveyed numbered 19 and 11 respectively.

As far as the staff strength of the respondents' institution is concerned, it is unevenly spread between the number of workers the various manufacturing companies have employed. Majority of the number of employees numbering less than 200 with the respondents' institutions is represented by 65.3 percent (109) of the total respondents whiles the number of staffs between

200 and 399 recorded 49 respondents representing 29.3 percent of the total respondents. The study recorded 9 respondents for the above 400 number of staffs of the respondents' institution representing 5.4 percent. These indications are presented in figure 4.3 below.

Staff Strength

120
100
80
60
40
20
< 200
200-399
Above 400

Figure 4.4: Staff Strength of Companies

Source: Fieldwork, 2023

4.3 Reliability Analysis

According to Sarantakos (2013), it is necessary to conduct a reliability test to ascertain whether a survey instrument is dependable to measure its intended measurement. The Cronbach alpha was then employed to measure the reliability for the items of the relevant variables. Cronbach alpha's reliability checks are conducted for the elements of the variables involved and are presented in table 4.4. With a value of 0.7 or above, the Cronbach alpha of a measurement item is considered adequate to measure its internal consistency. A value of 0.7 or higher is deemed acceptable when calculating an instrument's internal consistency. The reliabilities of these instruments are considered adequate as the values of Cronbach alpha falls above 0.7. Table 4.3 thus, provides the

overall internal reliability test for the survey instrument which shows that the reliability of the instrument is good.

Table 4.4: Results of Survey Instrument Reliability Test

| Cronbach's Alpha | Cronbach's Alpha Standardised Items | Based on | Number of Items |
|------------------|--|----------|-----------------|
| 0.8211 | 0.8211 | | 38 |
| | | | |

Source: Field Data, 2023.

Table 4.5: Variables Reliability

| Reliability | Items | Cronbach Alpha |
|---|-------|----------------|
| Information Technology | 5 | 0.8331 |
| Relationship Management (Supplier & Customer) | 8 | 0.7654 |
| Internal Integration | 3 | 0.7675 |
| Top Management | 5 | 0.7698 |
| Sustainable Supply Chain | 8 | 0.7857 |
| Supply Chain Challenges | 9 | 0.9221 |

Source: Fieldwork, 2023

4.4 Adopted Efficient Supply Chain Management Practices

This section of the chapter discusses the efficient supply chain management practices which cocoa companies have embraced and introduced. This section therefore aimed at evaluating to what degree were the ESCM activities adopted by companies in the cocoa industry, based on the main objective of this study. The respondents were presented with a variety of questions and replied on a scale of 1 to 5, 1 representing very small proportions and the 5 showing very large proportions. The mean and standard difference of variables used by the investigator shows in Table 4.6 to

Table 4.8 to show how far the respondents are pleased that the different ESCM practices have been implemented by their organizations. A 3-5 average suggests that the respondents'

organisation have taken the element in question to a large degree. The average 1-2 suggests that in a limited measure the organisation has acknowledged the factor in question.

The study examined the extent to which cocoa companies in Ghana have adopted information technology as an efficient tool in their supply chain management processes. Table 4.6 reveals that respondents agree to a large extent that their companies have a high level of information exchange with their major suppliers through the information network (mean = 3.72, SD = 1.11), and also have the foundation of a fast ordering system with main suppliers (mean = 3.59, SD = 1.08). These responses were complimented by respondents' agreement to a large extent that their organisations have stable procurement through the network system with their main suppliers (mean = 3.51, SD = 1.18). The respondents also agreed to a large extent that their organisations have high level of linkage with major customers through their information network (mean = 4.04, SD = 0.55). Responses again was in agreement to a large extent that the companies have high level of computerization for major customers' orders (mean = 3.90, SD = 0.76). Results clearly indicates that the cocoa companies have adopted information technology as an ESCM practice. The results reveals that information technology has enabled the cocoa companies to successfully manage the distribution of their products via strategic partnership. This result is in agreement with findings of O'Dwyer and Gilmore (2018) which revealed that e-planning as a system offers beneficial outcomes as it facilitates the efficient and effective sharing of ideas between customers and suppliers.

Table 4.6: Extent of Adoption of Information Technology as an ESCM Practice

| Information Technology | Mean | Std. dev |
|---|------|----------|
| We have high level of information exchange with major suppliers | 3.72 | 1.11 |

and through the information network

| We have the foundation of a fast ordering system with main | 3.59 | 1.08 |
|---|------|------|
| suppliers | | |
| We have stable procurement through the network system with our | 3.51 | 1.18 |
| main suppliers | | |
| We have high level of linkage with major customers through the | 4.04 | 0.55 |
| information network | | |
| We have high level of computerization for major customers' orders | 3.90 | 0.76 |
| | | |

Source: Fieldwork, 2023

Table 4.7 below reveals that the companies surveyed have adopted relationship management method as one of their ESCM practices. The study reveals that respondents agree to a large extent that their companies share a sense of fair play with their customers (mean = 3.01, SD = 1.34). Respondents also agree that their companies frequently interact with customers to set their reliability, responsiveness, and other standards (mean = 3.29, SD = 1.21). Responses also agree to a large extent that the companies frequent follow-up with their customers for quality/service feedback (mean = 3.34, SD = 1.35). Responses again indicate that respondents agree to a large extent that the companies frequently measure and evaluate their customer satisfaction (mean = 3.72, SD = 1.10).

Table 4.7: Extent of Adoption of Relationship Management as an ESCM PracticeRelationship ManagementMeanStd. dev

| Your organisation shares a sense of fair play with its customers | 3.01 | 1.34 |
|---|---------|------|
| Your organisation frequently interacts with customers to set its reliability, responsiveness, and other standards | 3.29 | 1.21 |
| Your organisation has frequent follow-up with its customers for | - | |
| quality/service feedback | 3.34 | 1.35 |
| Your organisation frequently measures and evaluates customer satisfaction | 3.72 | 1.10 |
| Your organisation frequently determine future customer expectation | ns 1.79 | 0.81 |
| Your organisation facilitates customers' ability to seek assistance | | |
| from it | 3.63 | 1.19 |
| Your organisation frequently evaluates the formal and informal | | |
| complaints of its customers | 3.72 | 1.12 |
| Your organisation periodically evaluates the importance of its relationship with its customers | 3.49 | 1.25 |
| | | |

Source: Fieldwork, 2023.

Responses from table 4.7 however revealed a disagreement to a large extent that cocoa companies are able to frequently determine future customer expectations (mean = 1.79, SD = 0.81) but agreed to a very large extent that they facilitate customers' ability to seek assistance from it (mean = 3.63, SD = 1.19), thereby indicating that the companies frequently evaluate the formal and informal complaints of their customers (mean = 3.72, SD = 1.12). Results further revealed that respondents agree to a large extent that their organisations periodically evaluate the importance of their relationships with their customers (mean = 3.49, SD = 1.25). Results therefore reveal that the cocoa companies have an instituted relationship management policy that facilitates the ESCM process. This result is in agreement with the findings of Daugherty *et al.* (2018) which noted that relationship management is a key factor in the facilitation process within the SC systems.

Table 4.8: Extent of Adoption of Internal Integration as an ESCM Practice

| Internal Integration | Mea | Std. dev |
|--|------|----------|
| | n | |
| Effective organizational internal integration is a high priority for | 3.97 | 0.86 |
| my organization | - | 10 |
| We have clearly established goals for inter-organizational | 3.96 | 0.88 |
| integration |) | |
| My organization recognizes the importance of working with other | 3.93 | 0.91 |
| organizations to achieve its mission | 3.93 | 0.91 |
| My organization considers the interests of other organizations in | 206 | 0.00 |
| its planning | 3.96 | 0.83 |
| My organization has committed adequate budget and resources to | | |
| inter-organizational integration | 3.89 | 0.81 |
| | | |
| My organization is willing to invest resources to accomplish | 3.91 | 0.76 |
| interorganizational integration goals | | |
| My organization can quickly form or modify partnerships as | 3.84 | 0.97 |
| requirements change | 2.0. | 0.5 / |
| My organization is flexible in adapting its processes and procedures | 3.92 | 0.91 |
| to better fit with other organizations | 3.92 | 0.91 |
| Our inter-organizational collaborations are effectively supported by | | 2 |
| collaborative planning tools and technologies | 3.84 | 0.88 |
| Our inter-organizational integration are supported by effective | 3.94 | 0.95 |
| communications tools and technologies | | |
| | | |

Source: Fieldwork, 2023.

Table 4.8 reveals that respondents agree to a large extent that effective organizational internal integration is a high priority for my organization (mean = 3.97, SD = 0.86), and they have clearly established goals for inter-organizational integration (mean = 3.96, SD = 0.88). Respondents further agreed to a very large extent that their organizations recognise the importance of working with other organizations to achieve its mission (mean = 3.93, SD = 0.91), and so they consider the

interests of other organizations in their planning (mean =3.96, SD = 0.83). By recording a corresponding average score of 3.89, 3.91, 3.84, 3.92, 3.84 and 3.94, respondents widely agreed respectively that their organizations have committed adequate budget and resources to interorganisational integration, their organisations demonstrate a willingness to allocate resources in order to achieve their goals of integrating with other organisations. Their organisations possess the ability to swiftly establish or adjust partnerships as the need arises. Organisations exhibit flexibility in modifying their processes and procedures to align more effectively with other organisations. The success of their collaborations between organisations is enhanced by the utilization of collaborative planning tools and technologies. Additionally, their interorganisational collaborations benefit from the implementation of effective communication tools and technologies. These findings resonate with that of Asamoah *et al.* (2021) who noted that building the needed resource capabilities within an organisation is necessary and so organisational internal integration systems serve as a valuable information resource which creates ESCM capabilities.

4.5 Factors Influencing the Adoption of Efficient SCM Practices

This section of the study also aims at identifying various factors with enormous influence on the adoption of ESCM practices. Table 4.11 and table 4.13 presents some factors and the extent to which the companies surveyed utilize them to influence their adoption of ESCM practices.

In response to top management as an influence on the adoption of efficient supply chain management practice, table 4.9 reveals that respondents agree to a large extent that top management continuously emphasized that the company must adopt efficient supply chain practices (mean = 3.63, SD = 1.19).

Table 4.9: Extent of Adoption of Top Management as an ESCM Practice

| Top Management | Mean | Std. dev |
|---|------|----------|
| Top management continuously emphasized that the company must adopt efficient supply chain practices | 3.63 | 1.19 |
| Top management play an effective role in coordination between supply chain partners | 3.72 | 1.12 |
| Top management makes consultations before selecting supply chain partners | 3.49 | 1.25 |
| Top management plays an effective role in conflict management on supply chain | 3.96 | 0.88 |
| Top management supports coordination among supply chain partners is high | 3.37 | 1.30 |

Source: Fieldwork, 2023.

The table 4.9 also revealed that respondents agree to a large extent that top management play an effective role in coordination between supply chain partners (mean = 3.72, SD = 1.12). Respondents further agreed to an extent that top management makes consultations before selecting supply chain partners (mean = 3.49, SD = 1.25). This is complimented by respondents' agreement to a large extent that top management plays an effective role in conflict management on supply chain (mean = 3.96, SD = 0.88). Respondents again agree to a large extent that top management supports coordination among supply chain partners is high (mean = 3.37, SD = 1.30). Results indicates that top management decisions had resulted in the adoption of various ESCM processes within the cocoa companies by significantly supporting various initiatives in harmonising the efficiency of the SC process. These findings were also noted by Kshetri (2018).

Table 4.10: Sustainable SC as a Factor Influencing ESCM Practices Adoption

| Sustainable Supply Chain | Mean | Std. dev |
|--|--------|----------|
| Company policies has resulted in significant improvement in terms o sales and market share. | f 3.86 | 1.06 |
| Company policies has resulted in significant reduction in terms of waste and its disposal costs. | 3.70 | 1.21 |
| Company policies has resulted in significant improvement in terms of resources management efficiency. | 4.11 | 1.04 |
| Company policies has resulted in to respond to and accommodate new products, new markets or new competitors | v 3.87 | 1.13 |
| Company policies has resulted in significant improvement in its image in the eyes of its customers. | e 3.83 | 0.79 |
| Company policies has resulted in significant improvement in relations with community stakeholders. | s 4.76 | 0. 69 |
| Company policies has resulted in significant reduction in consumption for hazardous/harmful/toxic materials. | n 3.92 | 1.42 |
| Company policies has resulted in significant reduction in energy consumption. | 3.79 | 1.25 |

Source: Fieldwork, 2023.

Table 4.10 shows that respondents agree to a large extent that supply chain flexibility of the various companies had been enhanced through their ability to respond to and accommodate demand variations, such as seasonality (mean = 3.86, SD = 1.06). Respondents also agreed to a large extent that their capacity to address and adapt to instances of poor production, such as mechanical malfunctions, had been enhanced (mean = 3.70, SD = 1.21) and that their capacity to address and adapt to instances of inadequate supplier output had likewise been enhanced (mean = 3.87, SD = 1.13). Results revealed that respondents agreed to a very high extent that their ability to respond to and accommodate the periods of poor supplier performance has been improved (mean = 4.11, SD = 1.04). Results clearly suggests that respondents agree to a large extent that the adopting of SCM practices has improved their ability to respond to and accommodate the

periods of poor delivery performance (mean = 3.87, SD = 1.13), and as well their ability to respond to and accommodate new products, new markets or new competitors has improved (mean = 3.83, SD = 0.79). Findings from the table above also indicates that respondents agree to a large extent that policies adopted by their companies had resulted in a significant improvement in their relations with community stakeholders (mean = 4.76, SD = 0.69). Again results revealed that respondents agree to a large extent that there is a significant reduction in the consumption of both hazardous/harmful/toxic materials (mean = 3.92, SD = 1.42), and energy (mean = 3.79, SD = 1.25) as a result of their company polities. Results were in consonance with findings from Yu *et al.* (2018) which noted that the adoption of SCM practices results in supply chain flexibility which allows the company to easily scale up to meet the demand of its customers. These findings are also not different from those of Delic and Eyers (2020).

4.6 The Impact of Efficient SCM Practices on Sustainable Supply Chain

According to Keith (2019), the regression analysis is an inferential statistical technique in which variables can be defined that either predict or explain the part of the total variance in the score of the dependent constructions. Table 4.11 above indicates the R2 value of 0.601, which demonstrates that the SCM practices are capable of explaining approximately 60% of the variance of the SC performance of manufacturing companies. As such, roughly 40 percent of the heterogeneity of the performance of business supply chain processes are clarified by other global influences, for which this study does not account for. The table also indicates statistically relevant relationships between ESCM practices and the sustainable supply chain result (Beta= 0.224, 0.401, 0.864; T = 2.909, 1.959, 1.002; p < 0.004, 0.000, 0.043). This finding show that the extent to which efficient supply chain management practices were adopted and introduced by the cocoa supply chain companies can be measured in the degree of business sustainable supply chain

process within cocoa companies. These findings were based on a preceding study by Nawi *et al*. (2016) which concluded that the quality of goods and services and increased business process results would benefit from ESCM practices, such as internal integration of systems, relationship management (both supplier and customer), and information technology. Specifically, ESCM practices comprising of information technology, relationship management, and internal integration practice respectively had a positive and significant impact on sustainable supply chain.

Table 4.11: Model Estimation of Efficient SCM Practices and Sustainable Supply Chain

| Independent | Dependent variable: Sustainable Supply Chain | | | | | | |
|-----------------------------|--|-------|-------|----------------------------|-------|--|--|
| variable: ESCM Practices | Standardiz <mark>ed</mark> Coefficients | T | Sig | Collinearity Statistics | | | |
| | Beta | | | Tol | VIF | | |
| Information | 0.224 | 2.909 | 0.004 | 0.673 | 1.584 | | |
| Technology | | | 1 | | | | |
| Relationship | 0.401 | 1.959 | 0.000 | 0.543 | 2.423 | | |
| Management | | 1 | | | - 5 | | |
| Internal Integration | 0.864 | 1.002 | 0.043 | 0.636 | 1.573 | | |

Model summary:

R = 0.775 Adjusted R2 = 0.601 F = 22.409

Tol=Tolerance VIF= Variance Inflation factor

Source: Fieldwork, 2023.

Table 4.12: Model Estimation of Top Management

| Independent | Dependent variable: Sustainable Supply Chain | | | | | | |
|----------------|--|-------|-------|-------------------------|-------|--|--|
| variable: | Standardized | T | Sig | Collinearity Statistics | | | |
| Top Management | Coefficients | | | | | | |
| | Beta | ANE ! | 10 | Tol | VIF | | |
| Top Management | 0.365 | 6.483 | 0.001 | 0.715 | 1.679 | | |
| Commitments | | | | | | | |

Model summary:

R = 0.517 Adjusted R2 = 0.691 F = 41.526

Tol=Tolerance VIF= Variance Inflation factor

Source: Fieldwork, 2023.

Results from table 4.12 reveals that top management commitment had been established to have positively and significantly impacted the sustainable supply chain (Beta = 0.365; T= 6.483; p = 0.001). This illustrates strong prediction of sustainable supply chain by the top management commitment. Table 4.12 also reveals approximately 69% (R2 = 0.691) variation with the sustainable supply chain. Results are consistent with findings of Hassan *et al.* (2017) which indicated that top management commitment are essential to the breadth and depth influence on sustainable supply chain among supply chain cocoa companies.

Results from table 4.13 reveals that top management moderating as a resource capability have been established to have positively and significantly impacted the relationship between ESCM practices and Sustainable SC thus, influencing the adoption and implementation of Efficient SCM practices (Beta = 0.472; T= 4.240; p = 0.000). This illustrates strong prediction of sustainable supply chain by the moderating factor. Table 4.13 also reveals approximately 72% ((R2 = 0.724) variation with the sustainable supply chain. Results are consistent with findings of Hassan *et al.* (2017) which indicated that the moderating factor of resource capability are essential to attaining a high level sustainable supply chain among supply chain companies.

Table 4.13: Model Estimation of the Moderating Factor of Top Management

| Independent | Dependent variable: Sustainable Supply Chain | | | | | |
|--------------------|--|-------|-------|------------|--------|--|
| variable: | Standardized | T | Sig | Collin | earity | |
| Moderating Factors | Coefficients | | | Statistics | | |
| | Beta | | | Tol | VIF | |
| TM * SCMP | 0.472 | 4.240 | 0.000 | 0.567 | 1.874 | |

Model summary:

R = 0.79 Adjusted R2 = 0.724 F = 12.763

Tol=Tolerance VIF= Variance Inflation factor

Source: Fieldwork, 2023.

4.7 Challenges Affecting the Adoption and Implementation of ESCM Practices

The study also sought to determine the challenges encountered by the respondents' companies in the adoption and implementation of efficient supply chain management practices. According to the results presented in table 4.14 below, respondents disagree to some extent that their companies lack professional expertise to implement ESCM practices (mean =2.36, SD = 1.17), respondents also disagree to some extent that their organisations lack frequent Efficient Supply Chain Management Practices training (mean=2.28, SD = 0.95). Results further show that respondents disagree to some extent that the cocoa companies had low level of cooperation among user departments (mean =2.92, SD =1.09). Again, respondents disagree to some extent that the companies had inadequate supplier relationship with their partners (mean =2.51, SD =1.07). Respondents further disagreed to some extent that their companies had poor information flow between them and their partners (mean =2.51, SD =1.15). Respondents disagree to some extent that their companies do not only concentrate on their core service (mean =2.31, SD = 1.23), respondents also disagree to some extent that their companies lack adequate storage capacities (mean=2.52, SD = 0.82). Results further show that respondents disagree to some extent that the cocoa companies lack adequate measures for monitoring and evaluation of ESCM practices to ensure Sustainable Supply Chain performance (mean =2.63, SD =0.67). Respondents however, agreed to some extent that rising logistics cost (Transportation, Material handling and procurement) was a challenge affecting the adoption of ESCM practices (mean =3.01, SD =1.21). These findings were not in agreement with the findings of Singh and Singh (2019) who noted that

there were a lot of challenges such as the lack of ESCM practice professionals to help various organisations to adopt and thereafter implement those practices. The findings of Singh and Singh were not different from those of Touboulic *et al.* (2020).

Table 4.14: Challenges Affecting the Adoption of Efficient SCM Practices

| Statements | Mean | Std. dev |
|---|------|----------|
| Our company lack professional expertise to implement ESCM practices | 2.36 | 1.17 |
| Our organization lacks frequent ESCM practices training | 2.28 | 0.95 |
| Our company has rising logistics cost (Transportation, Material handling and procurement) | 3.01 | 1.21 |
| The company has low level of cooperation among user departments | 2.92 | 1.09 |
| Inadequate Supplier relationship with our partners | 2.51 | 1.07 |
| Our company has poor Information flow between partners | 2.51 | 1.15 |
| We do not only concentrate on our core service | 2.31 | 1.23 |
| We lack adequate storage capacities | 2.52 | 0.82 |
| Inadequate measures for monitoring and evaluation of ESCM practices to ensure Sustainable Supply Chain Management | 2.63 | 0.67 |

Source: Fieldwork, 2023.

CHAPTER FIVE SUMMARY, CONCLUSION AND RECOMMENDATION

5.1 Introduction

The purpose of this study was to investigate the effects of effective supply chain management practices on the viability of the supply chain in the long term. Additionally, it aimed to analyse the influence of top management on this relationship. The study comprised four primary goals: firstly, to scrutinize the diverse effective practices in supply chain management within cocoa

enterprises in Ghana; secondly, to ascertain the influence of efficient supply chain management practices on the durability of the supply chain; thirdly, to ascertain the extent of involvement by top management in the association between ESCM practices and sustainable SC, and to determine the challenges with supply chain management practices. This chapter, thus, presents the summary of the findings pertaining to the aforementioned four objectives. Additionally, it elucidates the conclusions drawn and the recommendations formulated based on these findings. Furthermore, it offers suggestions on areas that warrant further research in relation to this concept. The findings of the preceding chapter are summarized in this chapter.

5.2 Summary of Findings

The study found that Ghana's cocoa companies adopted different efficient supply chain management practices in order to boost the sustainability of their supply chains. It also emerged from the report that the cocoa companies in Ghana followed the three efficient supply chain management practices: information technology, relationship management, and internal integration. Results showed that top management as a moderating variable have had a positive moderating effect on the relationship between efficient supply chain management practices and sustainable supply chain.

It was also evident from the report that the three supply chain management practices have had a positive effect on the outcome of the sustainable supply chain process. However, strategic supplier partnership and information sharing have had the greatest effects. A study of the relationship between efficient supply chain management practices and the sustainable supply chain of the business process has shown that practices can directly influence the sustainability of the business process. The study also found that organizations had no issues with the complexities that could emerge from the introduction and implementation of efficient supply chain management practices

which included lack professional expertise, lacks of frequent Efficient Supply Chain Management training, low level of cooperation among user departments, poor information flow between partners, adequate storage capacities, and inadequate measures for monitoring and evaluation of ESCM practices to ensure sustainable supply chain. However, rising logistics cost (Transportation, Material handling and procurement) has been a challenge. Thus, the results of this research point to the value of efficient supply chain management practices in the attainment of a sustainable supply chain process for both businesses and their suppliers/customers.

5.3 Conclusion

The study concludes that different efficient supply chain management practices were implemented by the cocoa companies that were surveyed. Practices also helped the organisations to boost the sustainability and efficiency of their activities. This is reinforced by the findings of the research that have shown that there is a causal effect between efficient supply chain management practices and the sustainability of the supply chain process. The study concluded that efficient supply chain management practices are very critical for improving the efficiency and performance of organisations and, as is understood, today's competition within the business world is fast changing from a manual systems of organizations to more powerful and competitive modelled systems. The study further showed that the businesses were in a position to deal with almost all the problems associated with the adoption of these practices. More and more companies are implementing efficient supply chain management practices in the quest of reducing logistics and transportation costs and achieving sustainable competitive advantages.

W SANE NO

5.4 Recommendation

The study has established that efficient supply chain management practices are very relevant in improving the sustainability of business processes for cocoa organisations. Both cocoa companies and other companies should adopt this idea because there are many advantages that will be gained. Organisations are encouraged to follow the methods that are followed by a very large degree because they can positively influence business process results and contribute to enhancement of the sustainability and efficiency in the company. Some of these activities include information technology, relationship management, and internal integration which have been considered effective within the cocoa industry. This study further suggests that in light of the challenges identified from literature which are likely to be encountered in the adoption and implementation of the efficient supply chain management practices, workshops and seminars should be organised for the workers to help enlightened them on the challenges and how they can be mitigated and also an oversight supervisory should be put in place to make sure that all measures for dealing with the challenges are carried out.

5.5 Suggestions for Further Research

It is a very diverse process so as the domain cannot be covered in only one research. Future research on efficient supply chain management practices implementations can be extended to include dimensions such as geographical proximity, cross functional collaboration, information system integration and dedicated supply chain integration that has been overlooked in this report. Future studies are suggested on the following aspects: the role of particular factors in developing supply chain systems and the consequences of such management practices for corporations and supply chain sustainability.

KNUST



REFERENCE

- Abbas, J., Mahmood, S., Ali, H., Ali Raza, M., Ali, G., Aman, J., ... and Nurunnabi, M. (2019). The effects of corporate social responsibility practices and environmental factors through a moderating role of social media marketing on sustainable performance of business firms. *Sustainability*, 11(12), pp. 3434.
- Abdallah, A. B., and Al-Ghwayeen, W. S. (2019). Green supply chain management and business performance: The mediating roles of environmental and operational performances. *Business Process Management Journal*, 26(2), pp. 489-512.
- Afum, E., Osei-Ahenkan, V. Y., Agyabeng-Mensah, Y., Owusu, J. A., Kusi, L. Y., and Ankomah, J. (2020). Green manufacturing practices and sustainable performance among Ghanaian manufacturing SMEs: the explanatory link of green supply chain integration. *Management of Environmental Quality: An International Journal*, 31(6), pp. 1457-1475.
- Alagidede, P., Mensah, J. O., and Ibrahim, M. (2018). Optimal deficit financing in a constrained fiscal space in Ghana. *African Development Review*, 30(3), pp. 291-303.
- Algozzine, B., and Hancock, D. (2017). *Doing case study research: A practical guide for beginning researchers*. Teachers College Press.
- Ali, M. H., Chung, L., Kumar, A., Zailani, S., and Tan, K. H. (2021). A sustainable Blockchain framework for the halal food supply chain: Lessons from Malaysia. *Technological Forecasting and Social Change*, 170, pp. 120870.
- Allaoui, H., Guo, Y., and Sarkis, J. (2019). Decision support for collaboration planning in sustainable supply chains. *Journal of Cleaner Production*, 229, pp. 761-774.

- Al-Sheyadi, A., Muyldermans, L., and Kauppi, K. (2019). The complementarity of green supply chain management practices and the impact on environmental performance. *Journal of environmental management*, 242, pp. 186-198.
- Alshurideh, M., Kurdi, B., Alzoubi, H., Obeidat, B., Hamadneh, S., and Ahmad, A. (2022). The influence of supply chain partners' integrations on organizational performance: The moderating role of trust. *Uncertain Supply Chain Management*, 10(4), pp. 1191-1202.
- Amagoh, F. (2016). Systems and complexity theories of organizations. *Global Encyclopedia of Public Administration, Public Policy, and Governance*.
- Aman, J., Abbas, J., Shi, G., Ain, N. U., and Gu, L. (2022). Community wellbeing under ChinaPakistan economic corridor: role of social, economic, cultural, and educational factors in improving residents' quality of life. *Frontiers in Psychology*, 12, pp. 816592.
- Amankwah-Amoah, J., Debrah, Y. A., and Nuertey, D. (2018). Institutional legitimacy, crossborder trade and institutional voids: Insights from the cocoa industry in Ghana. *Journal of Rural Studies*, 58, pp. 136-145.
- Ambulkar, S., Blackhurst, J., and Grawe, S. (2015). Firm's resilience to supply chain disruptions:

 Scale development and empirical examination. *Journal of operations management*, 33, pp. 111-122.
- Amoako, D. K., Zakuan, M. N., Okyere-Kwakye, E., and Tetteh, F. K. (2021). Effect of Training and Reward on Social Sustainability in Ghana's Cocoa Supply Chain: The Role of Green Buyer-Supplier Relationship. *Journal of International Food & Agribusiness Marketing*, pp. 1-32.

- Antoni, D., and Akbar, M. (2019). E-supply chain management value concept for the palm oil industry. *Jurnal Sistem Informasi*, 15(2), pp. 15-29.
- Arredondo, C. R., and Alfaro Tanco, J. A. (2021). Supply Chain Management: some reflections to improve its influence in business strategy. *Innovar*, *31*(81), pp. 7-19.
- Arshad Ali, A., Mahmood, A., Ikram, A., and Ahmad, A. (2020). Configuring the drivers and carriers of process innovation in manufacturing organizations. *Journal of Open Innovation: Technology, Market, and Complexity*, 6(4), pp. 154.
- Arunachalam, D., Kumar, N., and Kawalek, J. P. (2018). Understanding big data analytics capabilities in supply chain management: Unravelling the issues, challenges and implications for practice. *Transportation Research Part E: Logistics and Transportation Review*, 114, pp. 416-436.
- Arunmozhi, M., Kiran Kumar, R., and Srinivasa, B. A. (2021). Impact of COVID-19 on global supply chain management. In *Managing Supply Chain Risk and Disruptions: Post COVID-19* (pp. 1-18). Springer, Cham.
- Bag, S., Pretorius, J. H. C., Gupta, S., and Dwivedi, Y. K. (2021). Role of institutional pressures and resources in the adoption of big data analytics powered artificial intelligence, sustainable manufacturing practices and circular economy capabilities. *Technological Forecasting and Social Change*, 163, pp. 120420.
- Bag, S., Telukdarie, A., Pretorius, J. C., and Gupta, S. (2021). Industry 4.0 and supply chain sustainability: framework and future research directions. *Benchmarking: An International Journal*, 28(5), pp. 1410-1450.

- Bai, C., Quayson, M., and Sarkis, J. (2022). Analysis of Blockchain's enablers for improving sustainable supply chain transparency in Africa cocoa industry. *Journal of Cleaner Production*, 358, pp. 131896.
- Barber, K. D., Garza-Reyes, J. A., Kumar, V., and Abdi, M. R. (2017). The effect of supply chain management practices on supply chain and manufacturing firms' performance. *Journal of Manufacturing Technology Management*, 8(1), 19-28.
- Barykin, S. Y., Bochkarev, A. A., Dobronravin, E., and Sergeev, S. M. (2021). The place and role of digital twin in supply chain management. *Academy of Strategic Management Journal*, 20, pp. 1-19.
- Basheer, M., Siam, M., Awn, A., and Hassan, S. (2019). Exploring the role of TQM and supply chain practices for firm supply performance in the presence of information technology capabilities and supply chain technology adoption: A case of textile firms in Pakistan.

 *Uncertain Supply Chain Management, 7(2), pp. 275-288.
- Baumann, M., Moody, T. S., Smyth, M., and Wharry, S. (2020). A perspective on continuous flow chemistry in the pharmaceutical industry. *Organic Process Research and Development*, 24(10), pp. 1802-1813.
- Becker, H. S. (2017). Problems of inference and proof in participant observation. In *Research*Design (pp. 312-324). Routledge.
- Beier, G., Niehoff, S., and Xue, B. (2018). More sustainability in industry through industrial internet of things? *Applied sciences*, 8(2), pp. 219.
- Ben Abdelaziz, S. I., Saeed, M. A., and Benleulmi, A. Z. (2015). Social media effect on sustainable products purchase. In *Sustainability in Logistics and Supply Chain*

- Management: New Designs and Strategies. Proceedings of the Hamburg International Conference of Logistics (HICL), Vol. 21 (pp. 63-93). Berlin: epubli GmbH.
- Bernard, H. R. (2017). Research methods in anthropology: Qualitative and quantitative approaches. Rowman and Littlefield.
- Beske, P., Land, A., and Seuring, S. (2014). Sustainable supply chain management practices and dynamic capabilities in the food industry: A critical analysis of the literature. International journal of production economics, 152, pp. 131-143.
- Broman, G. I., and Robèrt, K. H. (2017). A framework for strategic sustainable development.

 *Journal of cleaner production, 140, pp. 17-31.
- Brunaud, B., Laínez-Aguirre, J. M., Pinto, J. M., and Grossmann, I. E. (2019). Inventory policies and safety stock optimization for supply chain planning. *AIChE journal*, 65(1), pp. 99112.
- Bryman, A. (2017). Quantitative and qualitative research: further reflections on their integration.

 In *Mixing methods: Qualitative and quantitative research* (pp. 57-78). Routledge.
- Cankaya, S. Y., and Sezen, B. (2018). Effects of green supply chain management practices on sustainability performance. *Journal of Manufacturing Technology Management*, 30(1), pp. 98-121.
- Cha, K. J., and Kim, Y. S. (2018). Critical success factors for mutual collaboration with suppliers in IT outsourcing industry: a case study of a top IT outsourcing company in Korea. Enterprise Information Systems, 12(1), pp. 76-95.
- Chege, S. M., and Wang, D. (2020). The influence of technology innovation on SME performance through environmental sustainability practices in Kenya. *Technology in Society*, 60, pp. 101210.

- Chen, J. H., and Wu, S. I. (2016). The impact of customer relationship management and internal marketing on business performance: a comparison of lodging industries. *Total Quality Management and Business Excellence*, 27(1-2), pp. 17-33.
- Chen, X., Sun, X., Yan, D., and Wen, D. (2020). Perceived sustainability and customer engagement in the online shopping environment: The rational and emotional perspectives.

 Sustainability, 12(7), pp. 2674.
- Choi, T. M., Cai, Y. J., and Shen, B. (2018). Sustainable fashion supply chain management: A system of systems analysis. *IEEE Transactions on Engineering Management*, 66(4), pp. 730-745.
- Creswell, J. W., and Creswell, J. D. (2017). *Research design: Qualitative, quantitative, and mixed methods approaches*. Sage publications.
- Danese, P., Lion, A., and Vinelli, A. (2019). Drivers and enablers of supplier sustainability practices: a survey-based analysis. *International Journal of Production Research*, *57*(7), pp. 2034-2056.
- De Propris, L., and Bailey, D. (2021). Pathways of regional transformation and Industry 4.0.

 *Regional Studies, 55(10-11), pp. 1617-1629.
- Di Fazio, S., and Modica, G. (2018). Historic rural landscapes: Sustainable planning strategies and action criteria. The Italian experience in the global and European context. *Sustainability*, *10*(11), pp. 3834.
- Dietrich, F., Ge, Y., Turgut, A., Louw, L., and Palm, D. (2021). Review and analysis of blockchain projects in supply chain management. *Procedia computer science*, 180, pp. 724-733.

- Dolgui, A., Ivanov, D., and Rozhkov, M. (2020). Does the ripple effect influence the bullwhip effect? An integrated analysis of structural and operational dynamics in the supply chain.

 International Journal of Production Research, 58(5), pp. 1285-1301.
- Dou, Y., Zhu, Q., and Sarkis, J. (2018). Green multi-tier supply chain management: An enabler investigation. *Journal of Purchasing and Supply Management*, 24(2), 95-107.
- Durso, F. T., Rawson, K. A., and Girotto, S. (2017). Comprehension and situation awareness. In *Situational Awareness* (pp. 77-108). Routledge.
- El Baz, J., Laguir, I., and Stekelorum, R. (2018). Logistics and supply chain management research in Africa: A systematic literature review and research agenda. *The International Journal of Logistics Management*, 30(1), pp. 8-38.
- Errassafi, M., Abbar, H., and Benabbou, Z. (2019). The mediating effect of internal integration on the relationship between supply chain integration and operational performance:

 Evidence from Moroccan manufacturing companies. *Journal of Industrial Engineering and Management*, 12(2), pp. 254-273.
- Esper, T. L., Castillo, V. E., Ren, K., Sodero, A., Wan, X., Croxton, K. L., ... and Goldsby, T. J. (2020). Everything Old is New Again: The Age of Consumer-Centric Supply Chain Management. *Journal of Business Logistics*, 41(4), pp. 286-293.
- Etikan, I., Musa, S. A., and Alkassim, R. S. (2016). Comparison of convenience sampling and purposive sampling. *American journal of theoretical and applied statistics*, 5(1), pp. 1-4.
- Faheem, M., Shah, S. B. H., Butt, R. A., Raza, B., Anwar, M., Ashraf, M. W., ... and Gungor, V.
 C. (2018). Smart grid communication and information technologies in the perspective of Industry 4.0: Opportunities and challenges. *Computer Science Review*, 30, pp. 1-30.

- Famiyeh, S., and Kwarteng, A. (2018). Implementation of environmental management practices in the Ghanaian mining and manufacturing supply chains. *International Journal of Productivity and Performance Management*, 67(7), pp. 1091-1112.
- Fischer, M., Imgrund, F., Janiesch, C., and Winkelmann, A. (2020). Strategy archetypes for digital transformation: Defining meta objectives using business process management. *Information and Management*, 57(5), pp. 103262.
- Fleming, J., and Zegwaard, K. E. (2018). Methodologies, Methods and Ethical Considerations for Conducting Research in Work-Integrated Learning. *International Journal of Work-Integrated Learning*, 19(3), pp. 205-213.
- Foo, P. Y., Lee, V. H., Tan, G. W. H., and Ooi, K. B. (2018). A gateway to realising sustainability performance via green supply chain management practices: A PLS–ANN approach. *Expert Systems with Applications*, 107, pp. 1-14.
- Frost, J. (2018). Difference between descriptive and inferential statistics. *Retrieved October*, 26, 2018.
- Gaudenzi, B., Mola, L., and Rossignoli, C. (2021). Hitting or missing the target: Resources and capabilities for alternative e-commerce pathways in the fashion industry. *Industrial Marketing Management*, 93, pp. 124-136.
- Giannakis, M., Spanaki, K., and Dubey, R. (2019). A cloud-based supply chain management system: effects on supply chain responsiveness. *Journal of Enterprise Information Management*, 32(4), pp. 585-607.
- Gil-Gomez, H., Guerola-Navarro, V., Oltra-Badenes, R., and Lozano-Quilis, J. A. (2020). Customer relationship management: digital transformation and sustainable business model innovation. *Economic research-Ekonomska istraživanja*, *33*(1), pp. 2733-2750.

- Glavee-Geo, R., Burki, U., and Buvik, A. (2020). Building trustworthy relationships with smallholder (small-scale) agro-commodity suppliers: Insights from the Ghana cocoa industry. *Journal of Macromarketing*, 40(1), pp. 110-127.
- Golgeci, I., and Gligor, D. M. (2017). The interplay between key marketing and supply chain management capabilities: the role of integrative mechanisms. *Journal of Business and Industrial Marketing*, 32(3), pp. 472-483.
- Golicic, S. L., and Smith, C. D. (2013). A meta-analysis of environmentally sustainable supply chain management practices and firm performance. *Journal of supply chain management*, 49(2), pp. 78-95.
- Gouda, S. K., and Saranga, H. (2018). Sustainable supply chains for supply chain sustainability: impact of sustainability efforts on supply chain risk. *International Journal of Production Research*, 56(17), pp. 5820-5835.
- Govindan, K., and Bouzon, M. (2018). From a literature review to a multi-perspective framework for reverse logistics barriers and drivers. *Journal of cleaner production*, 187, 318-337.
- Govindan, K., Cheng, T. E., Mishra, N., and Shukla, N. (2018). Big data analytics and application for logistics and supply chain management. *Transportation Research Part E: Logistics and Transportation Review*, 114, pp. pp. 343-349.
- Govindan, K., Seuring, S., Zhu, Q., and Azevedo, S. G. (2016). Accelerating the transition towards sustainability dynamics into supply chain relationship management and governance structures. *Journal of cleaner production*, *112*, pp. 1813-1823.
- Gružauskas, V., and Vilkas, M. (2017). Managing capabilities for supply chain resilience through it integration. *Economics and Business*, *31*(1), pp. 30-43.

- Gružauskas, V., Baskutis, S., and Navickas, V. (2018). Minimizing the trade-off between sustainability and cost effective performance by using autonomous vehicles. *Journal of Cleaner Production*, 184, pp. 709-717.
- Gupta, S., Chen, H., Hazen, B. T., Kaur, S., and Gonzalez, E. D. S. (2019). Circular economy and big data analytics: A stakeholder perspective. *Technological Forecasting and Social Change*, 144, pp. 466-474.
- Haddouch, H., Beidouri, Z., and El Oumami, M. (2019). Supply chain management: A review of approaches, practices and impact on performance. *International Journal of Supply Chain Management*, 8(6), pp. 1-13.
- Haiyun, C., Zhixiong, H., Yüksel, S., and Dinçer, H. (2021). Analysis of the innovation strategies for green supply chain management in the energy industry using the QFD-based hybrid interval valued intuitionistic fuzzy decision approach. *Renewable and Sustainable Energy Reviews*, 143, pp. 110844.
- Hanelt, A., Bohnsack, R., Marz, D., and Antunes Marante, C. (2021). A systematic review of the literature on digital transformation: Insights and implications for strategy and organizational change. *Journal of Management Studies*, 58(5), pp. 1159-1197.
- Herczeg, G., Akkerman, R., and Hauschild, M. Z. (2018). Supply chain collaboration in industrial symbiosis networks. *Journal of Cleaner Production*, *171*, pp. 1058-1067.
- Herold, D. M. (2018). Demystifying the link between institutional theory and stakeholder theory in sustainability reporting. *Economics, Management and Sustainability*, 3(2), pp. 6-19.
- Herrera, J., and de las Heras-Rosas, C. (2020). Economic, non-economic and critical factors for the sustainability of family firms. *Journal of Open Innovation: Technology, Market, and Complexity*, 6(4), pp. 119.

- Ho, A. D., and Yu, C. C. (2015). Descriptive statistics for modern test score distributions: Skewness, kurtosis, discreteness, and ceiling effects. *Educational and psychological measurement*, 75(3), pp. 365-388.
- Hohenstein, N. O., Feisel, E., Hartmann, E., and Giunipero, L. (2015). Research on the phenomenon of supply chain resilience: a systematic review and paths for further investigation. *International journal of physical distribution and logistics management*, 45(1/2), pp. 90-117.
- Hong, J., Zhang, Y., and Ding, M. (2018). Sustainable supply chain management practices, supply chain dynamic capabilities, and enterprise performance. *Journal of cleaner production*, 172, pp. 3508-3519.
- Hosseini, S., Ivanov, D., and Dolgui, A. (2019). Review of quantitative methods for supply chain resilience analysis. *Transportation research part E: logistics and transportation review*, 125, pp. 285-307.
- Huo, B., Zhang, C., and Zhao, X. (2015). The effect of IT and relationship commitment on supply chain coordination: A contingency and configuration approach. *Information and Management*, 52(6), pp. 728-740.
- Hussain, T., Abbas, J., Wei, Z., and Nurunnabi, M. (2019). The effect of sustainable urban planning and slum disamenity on the value of neighboring residential property:

 Application of the hedonic pricing model in rent price appraisal. *Sustainability*, *11*(4), pp. 1144.
- Isnaini, D. B. Y., Nurhaida, T., and Pratama, I. (2020). Moderating effect of supply chain dynamic capabilities on the relationship of sustainable supply chain management practices and

- organizational sustainable performance: A study on the restaurant industry in Indonesia. *International Journal of Supply Chain Management (IJSCM)*, 9(1), pp. 97-105.
- Jaradat, R., Adams, F., Abutabenjeh, S., and Keating, C. (2017). The complementary perspective of systems in collaboration, integration, and logistics: a value-chain based paradigm of supply chain management. *Systems*, 5(4), pp. 50.
- Jena, S. K., and Ghadge, A. (2021). An integrated supply chain–human resource management approach for improved supply chain performance. *The International Journal of Logistics Management*, 32(3), pp. 918-941.
- Ju, Y., Hou, H., and Yang, J. (2021). Integration quality, value co-creation and resilience in logistics service supply chains: moderating role of digital technology. *Industrial management and data systems*, 121(2), pp. 364-380.
- Kadam, R. A. (2017). Informed consent process: a step further towards making it meaningful! *Perspectives in clinical research*, 8(3), pp. 107.
- Kano, L., Tsang, E. W., and Yeung, H. W. C. (2020). Global value chains: A review of the multidisciplinary literature. *Journal of international business studies*, 51(4), pp. 577-622.
- Karmaker, C. L., and Ahmed, T. (2020). Modeling performance indicators of resilient pharmaceutical supply chain. *Modern Supply Chain Research and Applications*, 2(3), pp. 179-205.
- Khanuja, A., and Jain, R. K. (2019). Supply chain integration: a review of enablers, dimensions and performance. *Benchmarking: An international journal*, 27(1), pp. 264-301.
- Kittipanya-Ngam, P., and Tan, K. H. (2020). A framework for food supply chain digitalization: lessons from Thailand. *Production Planning and Control*, *31*(2-3), pp. 158-172.

- Klobučar, T., and Erjavec, J. (2019). Continuous Supplier Relationship Management Framework. *Poslovna izvrsnost*, 13(2), pp. 87-109.
- Köksal, D., Strähle, J., Müller, M., and Freise, M. (2017). Social sustainable supply chain management in the textile and apparel industry—A literature review. *Sustainability*, 9(1), pp. 100.
- Kot, S. (2018). Sustainable supply chain management in small and medium enterprises. *Sustainability*, 10(4), pp. 1143.
- Kumar, A., Mangla, S. K., Luthra, S., and Ishizaka, A. (2019). Evaluating the human resource related soft dimensions in green supply chain management implementation. *Production Planning and Control*, 30(9), pp. 699-715.
- Kumar, A., Shrivastav, S. K., and Oberoi, S. S. (2021). Application of analytics in supply chain management from industry and academic perspective. *FIIB Business Review*, 23197145211028041.
- Kusi-Sarpong, S., Gupta, H., and Sarkis, J. (2019). A supply chain sustainability innovation framework and evaluation methodology. *International Journal of Production Research*, 57(7), pp. 1990-2008.
- Lacomme, P., Moukrim, A., Quilliot, A., and Vinot, M. (2018). Supply chain optimisation with both production and transportation integration: multiple vehicles for a single perishable product. *International Journal of Production Research*, 56(12), pp. 4313-4336.
- Lee, R. (2021). The effect of supply chain management strategy on operational and financial performance. *Sustainability*, *13*(9), pp. 5138.

- Li, J., Anser, M. K., Tabash, M. I., Nassani, A. A., Haffar, M., and Zaman, K. (2021). Technology-and logistics-induced carbon emissions obstructing the Green supply chain management agenda: Evidence from 101 countries. *International Journal of Logistics Research and Applications*, pp. 1-25.
- Li, X., Abbas, J., Dongling, W., Baig, N. U. A., and Zhang, R. (2022). From cultural tourism to social entrepreneurship: Role of social value creation for environmental sustainability. *Frontiers in Psychology*, 13, pp. 925768.
- Liang, S., Rajora, M., Liu, X., Yue, C., Zou, P., and Wang, L. (2018). Intelligent manufacturing systems: a review. *International Journal of Mechanical Engineering and Robotics Research*, 7(3), pp. 324-330.
- Liu, C. L., and Lee, M. Y. (2018). Integration, supply chain resilience, and service performance in third-party logistics providers. *The international journal of logistics management*, 29(1), pp. 5-21.
- Longo, F., Nicoletti, L., Padovano, A., d'Atri, G., and Forte, M. (2019). Blockchain-enabled supply chain: An experimental study. *Computers and Industrial Engineering*, *136*, pp. 57-69.
- Lv, Z., Li, X., Wang, W., Zhang, B., Hu, J., and Feng, S. (2018). Government affairs service platform for smart city. *Future Generation Computer Systems*, 81, pp. 443-451.
- MacCarthy, B. L., Blome, C., Olhager, J., Srai, J. S., and Zhao, X. (2016). Supply chain evolution—theory, concepts and science. *International Journal of Operations and Production Management*, 36(12), pp. 1696-1718.

- Madhani, P. M. (2019). Strategic supply chain management for enhancing competitive advantages: developing business value added framework. *International Journal of Value Chain Management*, 10(4), pp. 316-338.
- Malak-Rawlikowska, A., Majewski, E., Was, A., Borgen, S. O., Csillag, P., Donati, M., ... and Wavresky, P. (2019). Measuring the economic, environmental, and social sustainability of short food supply chains. *Sustainability*, *11*(15), pp. 4004.
- Mancheri, N. A., Sprecher, B., Bailey, G., Ge, J., and Tukker, A. (2019). Effect of Chinese policies on rare earth supply chain resilience. *Resources, Conservation and Recycling*, *142*, pp. 101-112.
- Mangla, S. K., Luthra, S., Mishra, N., Singh, A., Rana, N. P., Dora, M., and Dwivedi, Y. (2018).

 Barriers to effective circular supply chain management in a developing country context.

 Production Planning and Control, 29(6), pp. 551-569.
- Mani, V., Gunasekaran, A., and Delgado, C. (2018). Enhancing supply chain performance through supplier social sustainability: An emerging economy perspective. *International Journal of Production Economics*, 195, pp. 259-272.
- Mastos, T. D., Nizamis, A., Vafeiadis, T., Alexopoulos, N., Ntinas, C., Gkortzis, D., ... and Tzovaras, D. (2020). Industry 4.0 sustainable supply chains: An application of an IoT enabled scrap metal management solution. *Journal of cleaner production*, 269, pp. 122377.
- Meixell, M. J., and Luoma, P. (2015). Stakeholder pressure in sustainable supply chain management: A systematic review. *International Journal of Physical Distribution and Logistics Management*, 45(1/2), pp. 69-89.

- Melander, L., and Pazirandeh, A. (2019). Collaboration beyond the supply network for green innovation: insight from 11 cases. *Supply Chain Management: An International Journal*, 24(4), pp. 509-523.
- Mena, J. A., Hult, G. T. M., Ferrell, O. C., and Zhang, Y. (2019). Competing assessments of market-driven, sustainability-centered, and stakeholder-focused approaches to the customer-brand relationships and performance. *Journal of Business Research*, 95, pp. 531-543.
- Min, S., Zacharia, Z. G., and Smith, C. D. (2019). Defining supply chain management: in the past, present, and future. *Journal of Business Logistics*, 40(1), pp. 44-55.
- Min, S., Zacharia, Z. G., and Smith, C. D. (2019). Defining supply chain management: in the past, present, and future. *Journal of Business Logistics*, 40(1), pp. 44-55.
- Mohajan, H. (2017). Two criteria for good measurements in research: Validity and reliability. *Annals of Spiru Haret University Economics Series*, (4), pp. 59-82.
- Morais, D. O. C., and Barbieri, J. C. (2022). Supply Chain Social Sustainability: Unveiling Focal Firm's Archetypes under the Lens of Stakeholder and Contingency Theory. Sustainability, 14(3), pp. 1185.
- Mzembe, A. N., Lindgreen, A., Maon, F., and Vanhamme, J. (2016). Investigating the drivers of corporate social responsibility in the global tea supply chain: A case study of Eastern Produce Limited in Malawi. *Corporate Social Responsibility and Environmental Management*, 23(3), pp. 165-178.
- Neuman, W. L. (2014). Social research methods: Qualitative and quantitative approaches:

 Pearson new international edition. Pearson Education Limited.

- Nimeh, H. A., Abdallah, A. B., and Sweis, R. (2018). Lean supply chain management practices and performance: empirical evidence from manufacturing companies. International Journal of Supply Chain Management, 7(1), pp. 1-15.
- Norman, A., and Naslund, D. (2019). Supply chain incentive alignment: The gap between perceived importance and actual practice. *Operations and Supply Chain Management: An International Journal*, 12(3), pp. 129-142.
- Oh, S., Ryu, Y. U., and Yang, H. (2019). Interaction effects between supply chain capabilities and information technology on firm performance. *Information Technology and Management*, 20, pp. 91-106.
- Omoregie, O. K., Addae, J. A., Coffie, S., Ampong, G. O. A., and Ofori, K. S. (2019). Factors influencing consumer loyalty: evidence from the Ghanaian retail banking industry. *International Journal of Bank Marketing*, 37(3), pp. 798-820.
- Pacheco, P., Schoneveld, G., Dermawan, A., Komarudin, H., and Djama, M. (2020). Governing sustainable palm oil supply: Disconnects, complementarities, and antagonisms between state regulations and private standards. *Regulation and Governance*, 14(3), pp. 568-598.
- Pananond, P., Gereffi, G., and Pedersen, T. (2020). An integrative typology of global strategy and global value chains: The management and organization of cross-border activities. *Global Strategy Journal*, 10(3), pp. 421-443.
- Papalexi, M., Bamford, D., and Breen, L. (2020). Key sources of operational inefficiency in the pharmaceutical supply chain. *Supply Chain Management: An International Journal*, 25(6), pp. 617-635.

- Pellegrini, C., Annunziata, E., Rizzi, F., and Frey, M. (2019). The role of networks and sustainable intrapreneurship as interactive drivers catalyzing the adoption of sustainable innovation.

 *Corporate Social Responsibility and Environmental Management, 26(5), pp. 1026-1048.
- Perboli, G., Musso, S., and Rosano, M. (2018). Blockchain in logistics and supply chain: A lean approach for designing real-world use cases. *Ieee Access*, 6, pp. 62018-62028.
- Piprani, A. Z., Mohezar, S., and Jaafar, N. I. (2020). Supply chain integration and supply chain performance: The mediating role of supply chain resilience. *International Journal of Supply Chain Management*, 9(3), pp. 58-73.
- Piya, S., Shamsuzzoha, A., Khadem, M., and Al-Kindi, M. (2017). Supply chain complexity drivers and solution methods. *International Journal of Supply Chain Management*, 6(4), pp. 43-50.
- Prajogo, D., Toy, J., Bhattacharya, A., Oke, A., and Cheng, T. C. E. (2018). The relationships between information management, process management and operational performance:

 Internal and external contexts. *International Journal of Production Economics*, 199, pp. 95-103.
- Pyykkö, H., Suoheimo, M., and Walter, S. (2021). Approaching Sustainability Transition in Supply Chains as a Wicked Problem: Systematic Literature Review in Light of the Evolved Double Diamond Design Process Model. *Processes*, 9(12), pp. 2135.
- Quynh, D. V. X., and Huy, N. H. (2018). Supply chain management practices, competitive advantages and firm performance: A case of small and medium enterprises (SMEs) in Vietnam. *Journal of modern accounting and auditing*, 14(3), pp. 136-146.
- Raj, A., Biswas, I., and Srivastava, S. K. (2018). Designing supply contracts for the sustainable supply chain using game theory. *Journal of cleaner production*, *185*, pp. 275-284.

- Rashid, Y., Rashid, A., Warraich, M. A., Sabir, S. S., and Waseem, A. (2019). Case study method:

 A step-by-step guide for business researchers. *International journal of qualitative methods*, 18, 1609406919862424.
- Rosati, F., and Faria, L. G. D. (2019). Business contribution to the Sustainable Development Agenda: Organizational factors related to early adoption of SDG reporting. *Corporate Social Responsibility and Environmental Management*, 26(3), pp. 588-597.
- Roy, V., Schoenherr, T., and Charan, P. (2018). The thematic landscape of literature in sustainable supply chain management (SSCM): A review of the principal facets in SSCM development. *International Journal of Operations and Production Management*, 38(4), pp. 1091-1124.
- Saeed, M. A., and Kersten, W. (2019). Drivers of sustainable supply chain management: identification and classification. *Sustainability*, *11*(4), pp. 1137.
- Saeed, M. A., Waseek, I., and Kersten, W. (2017). Literature review of drivers of sustainable supply chain management. In *Proceedings of the Hamburg international Conference of Logistics (HICL)* (pp. 159-184). epubli.
- Sajjad, A., Eweje, G., and Tappin, D. (2020). Managerial perspectives on drivers for and barriers to sustainable supply chain management implementation: Evidence from New Zealand.

 *Business Strategy and the Environment, 29(2), pp. 592-604.
- Sajjad, A., Eweje, G., and Tappin, D. (2020). Managerial perspectives on drivers for and barriers to sustainable supply chain management implementation: Evidence from New Zealand.

 *Business Strategy and the Environment, 29(2), pp. 592-604.
- Sanderson, J., Lonsdale, C., Mannion, R., and Matharu, T. (2015). Theories about procurement and supply chain management. In *Towards a framework for enhancing procurement and*

- supply chain management practice in the NHS: lessons for managers and clinicians from a synthesis of the theoretical and empirical literature. NIHR Journals Library.
- Schniederjans, D. G., Curado, C., and Khalajhedayati, M. (2020). Supply chain digitisation trends:

 An integration of knowledge management. *International Journal of Production Economics*, 220, pp. 107439.
- Schrettle, S., Hinz, A., Scherrer-Rathje, M., and Friedli, T. (2014). Turning sustainability into action: Explaining firms' sustainability efforts and their impact on firm performance.

 International Journal of Production Economics, 147, pp. 73-84.
- Schultz, F. C., Everding, S., and Pies, I. (2021). Circular supply chain governance: A qualitative empirical study of the European polyure than eindustry to facilitate functional circular supply chain management. *Journal of Cleaner Production*, 317, pp. 128445.
- Silverman, B. W. (2018). *Density estimation for statistics and data analysis*. Routledge.
- Singh, P. K., Sharma, S. K., Samuel, C., and Verma, S. (2017). Supplier relationship management and selection strategies—A literature review [Conference presentation]. In 4th International Conference on Industrial Engineering, Surat, India.
- Sundram, V. P. K., Ibrahim, A. R., and Govindaraju, V. C. (2011). Supply chain management practices in the electronics industry in Malaysia: Consequences for supply chain performance. *Benchmarking: An International Journal*, 18(6), pp. 834-855.
- Suryanto, T., Haseeb, M., and Hartani, N. H. (2018). The correlates of developing green supply chain management practices: Firms level analysis in Malaysia. *International Journal of Supply Chain Management*, 7(5), pp. 316.

- Suryanto, T., Haseeb, M., and Hartani, N. H. (2018). The correlates of developing green supply chain management practices: Firms level analysis in Malaysia. *International Journal of Supply Chain Management*, 7(5), pp. 316.
 - Taherdoost, H. (2017). Determining sample size; how to calculate survey sample size. *International Journal of Economics and Management Systems*, 2, pp. 237-239.
- Tarigan, Z. J. H., Siagian, H., and Jie, F. (2020). The role of top management commitment to enhancing the competitive advantage through ERP integration and purchasing strategy.

 International Journal of Enterprise Information Systems (IJEIS), 16(1), pp. 53-68.
- Tarigan, Z., Mochtar, J., Basana, S., and Siagian, H. (2021). The effect of competency management on organizational performance through supply chain integration and quality.

 *Uncertain Supply Chain Management, 9(2), pp. 283-294.
- Teece, D. J. (1982). Towards an economic theory of the multiproduct firm. *Journal of Economic Behavior and Organization*, 3(1), pp. 39-63.
- Testa, F., Boiral, O., and Iraldo, F. (2018). Internalization of environmental practices and institutional complexity: Can stakeholders pressures encourage greenwashing? *Journal of Business Ethics*, 147(2), pp. 287-307.
- Thöni, A., and Tjoa, A. M. (2017). Information technology for sustainable supply chain management: a literature survey. *Enterprise Information Systems*, 11(6), pp. 828-858.
- Tirkolaee, E. B., Goli, A., Ghasemi, P., and Goodarzian, F. (2022). Designing a sustainable closed-loop supply chain network of face masks during the COVID-19 pandemic: Paretobased algorithms. *Journal of Cleaner Production*, *333*, pp. 130056.

- Tseng, M. L., Islam, M. S., Karia, N., Fauzi, F. A., and Afrin, S. (2019). A literature review on green supply chain management: Trends and future challenges. *Resources, Conservation and Recycling*, *141*, pp. 145-162.
- Tsing, A. (2016). What is emerging? Supply chains and the remaking of Asia. *The Professional Geographer*, 68(2), pp. 330-337.
- Valdez-Juárez, L. E., Gallardo-Vázquez, D., and Ramos-Escobar, E. A. (2018). CSR and the Supply Chain: Effects on the Results of SMEs. *Sustainability*, *10*(7), pp. 2356.
- Van Thiel, S. (2014). Research methods in public administration and public management: An introduction. Routledge.
- Varadarajan, R. (2020). Customer information resources advantage, marketing strategy and business performance: A market resources based view. *Industrial Marketing Management*, 89, pp. 89-97.
- Vaske, J. J., Beaman, J., and Sponarski, C. C. (2017). Rethinking internal consistency in Cronbach's alpha. *Leisure sciences*, *39*(2), pp. 163-173.
- Vigneri, M., and Kolavalli, S. (2017). Growth through pricing policy: The case of cocoa in Ghana. *Background paper for UNCTAD-FAO Commodities and Development Report*.
- Vledder, M., Friedman, J., Sjöblom, M., Brown, T., and Yadav, P. (2019). Improving supply chain for essential drugs in low-income countries: results from a large scale randomized experiment in Zambia. *Health Systems and Reform*, 5(2), pp. 158-177.

- Wijethilake, C., and Lama, T. (2019). Sustainability core values and sustainability risk management: Moderating effects of top management commitment and stakeholder pressure. *Business Strategy and the Environment*, 28(1), pp. 143-154.
- Wikhamn, W. (2019). Innovation, sustainable HRM and customer satisfaction. *International Journal of Hospitality Management*, 76, 102-110.
- Wilhelm, M. M., Blome, C., Bhakoo, V., and Paulraj, A. (2016). Sustainability in multi-tier supply chains: Understanding the double agency role of the first-tier supplier. *Journal of operations management*, 41, pp. 42-60.
- Xu, J., Feng, Q., Lv, C., and Huang, Q. (2019). Low-carbon electricity generation—based dynamic equilibrium strategy for carbon dioxide emissions reduction in the coal-fired power enterprise. *Environmental Science and Pollution Research*, 26(36), pp. 36732-36753.
- Yu, W., Chavez, R., Feng, M., Wong, C. Y., and Fynes, B. (2020). Green human resource management and environmental cooperation: An ability-motivation-opportunity and contingency perspective. *International Journal of Production Economics*, 219, pp. 224235.
- Zaid, A. A., Jaaron, A. A., and Bon, A. T. (2018). The impact of green human resource management and green supply chain management practices on sustainable performance:

 An empirical study. *Journal of cleaner production*, 204, pp. 965-979.
- Zhang, A., Tay, H. L., Alvi, M. F., Wang, J. X., and Gong, Y. (2022). Carbon neutrality drivers and implications for firm performance and supply chain management. *Business Strategy and the Environment*, pp. 1-15.

- Zhang, J., Kang, L., Li, H., Ballesteros-Pérez, P., Skitmore, M., and Zuo, J. (2020). The impact of environmental regulations on urban Green innovation efficiency: The case of Xi'an. Sustainable Cities and Society, 57, pp. 102123.
- Zhu, C., Guo, X., and Zou, S. (2022). Impact of information and communications technology alignment on supply chain performance in the Industry 4.0 era: Mediation effect of supply chain integration. *Journal of Industrial and Production Engineering*, 39(7), 505-520.
- Zhu, Q., Sarkis, J., and Lai, K. H. (2013). Institutional-based antecedents and performance outcomes of internal and external green supply chain management practices. Journal of Purchasing and Supply Management, 19(2), pp. 106-117.
- Ziggers, G. W., and Henseler, J. (2016). The reinforcing effect of a firm's customer orientation and supply-base orientation on performance. *Industrial marketing management*, *52*, pp. 18-26.
- Zimon, D., Tyan, J., and Sroufe, R. (2020). Drivers of sustainable supply chain management:

 Practices to alignment with un sustainable development goals. *International Journal for Quality Research*, 14(1), pp. 219-236.

RISADS WY SANE

RESEARCH QUESTIONNAIRE

My name is Mather Darkwa, a student of Kwame Nkrumah University of Science and Technology, conducting a research on the topic "EVALUATING TEHE EFFECT OF EFFICIENT SUPPLY CHAIN MANAGEMENT ON SUSTAINABLE SUPPLY CHAIN:

A MODERATING EFFECT OF TOP MANAGEMENT." The purpose of the study is purely academic and all information provided will be treated with strict confidentiality.

SECTION A: DEMOGRAPHIC OF RESPONDENTS

Please understudy the demographic information and tick $[\sqrt{\ }]$ appropriately as applicable.

| 1) Please indicate the age bracket yo a) 20-30 years | u fall i <mark>n as below. []</mark> |
|--|--|
| b) 31-40 years | |
| c) 41-50 years | |
| d) 51-60 years | 11 |
| e) Above 60 years | |
| 6 | ELECTION |
| 2) Please indicate your gender | E TOSSER |
| a) Male | |
| b) Female | |
| | |
| 3) Your level of education | |
| a) Professional certificate | |
| b) Н. N. D | |
| c) Undergraduate. | [] BAD |
| d) Postgraduate. | SANE NO |
| e) No formal education | |
| 4) How long have you been working | with the organisation? |
| a) Under 5 years | |

- b) 5 10 years []
- c) 11 15 years []
- d) 16 20 years []
- e) Over 20 years

KNUST

- 5) What is your designation?
 - a) Senior Manager []
 - b) Manager []
 - c) Supervisor []
 - d) Senior staff []
 - e) Junior staff []
- 6) Number of employees in your company is?
 - a) Less than 50 []
 - b) 50 100 []
 - c) 101 250 []
 - d) Above 250 []

SECTION B: EFFICIENT SUPPLY CHAIN MANAGEMENT

With regards to Efficient Supply Chain Management, please tick $[\ \ \ \]$ the appropriate number to indicate the extent to which you agree or disagree with each statement. The item scales are fivepoint Likert type scale with 1 = Strongly Disagree (SD), 2 = Disagree (D), 3 = Neutral (N), 4 = Agree (A), 5 = Strongly Agree (SA).

| Efficient Supply Chain management Practices | SD | D | N | A | SA |
|---|----|---|-----|-------|----|
| | 1 | 2 | 3 | 4 | 5 |
| Information Technology | | T | 100 | | |
| We have high level of information exchange with major supplier through the information network |) | ı | | | |
| We have the foundation of a fast ordering system with main supplier | | | | | |
| We have stable procurement through the network system with our main suppliers | | | | | |
| We have high level of linkage with major customers through the information network | | | | | |
| We have high level of computerization for major customers' orders |) | | | | |
| Relationship Management (Supplier and Customer) | 1 | | | | |
| Your organisation shares a sense of fair play with its customers | 3 | 2 | 2 | 5 | 1 |
| Your organisation frequently interacts with customers to set its reliability, responsiveness, and other standards | 8 | S | 3 | V. | |
| Your organisation has frequent follow-up with its customers for quality/service feedback | | | |) | |
| Your organisation frequently measures and evaluates customer satisfaction | | | | | A |
| Your organisation frequently determine future customer expectations | | | 3 | 1/1/5 | 1 |
| Your organisation facilitates customers' ability to seek assistance from it | 18 | | | | |
| Your organisation frequently evaluates the formal and informal complaints of its customers | | | | | |
| Your organisation periodically evaluates the importance of its relationship with its customers | | | | | |

| Internal Integration | | | | | |
|--|---|---|---|---|---|
| Effective organizational internal integration is a high priority for my organization | | | | | |
| We have clearly established goals for inter-organizational integration | - | T | | | |
| My organization recognizes the importance of working with other organizations to achieve its mission | 9 | 1 | | | |
| My organization considers the interests of other organizations in its planning | | | | | |
| My organization has committed adequate budget and resources to inter-organizational integration | | | | | |
| My organization is willing to invest resources to accomplish inter-organizational integration goals | | | | | |
| My organization can quickly form or modify partnerships as requirements change | į | | | | |
| My organization is flexible in adapting its processes and procedures to better fit with other organizations | | | | | |
| Our inter-organizational collaborations are effectively supported by collaborative planning tools and technologies | 1 | | | | 5 |
| Our inter-organizational integration are supported by effective communications tools and technologies | 3 | | 2 | 7 | 7 |

SECTION C: TOP MANAGEMENT

With regards to Top Management, please tick $\lceil \sqrt{\rceil}$ appropriately the number that reflects your firm's present condition. The item scales are five-point Likert type scale with 1 = Strongly Disagree, 2 = Disagree, 3 = Neutral, 4 = Agree, 5 = Strongly Agree.

| Top Management | 1 | 2 | 3 | 4 | 5 | |
|----------------|---|---|---|---|---|--|
|----------------|---|---|---|---|---|--|

| Top management continuously emphasized that the company must | | | |
|---|---|--|--|
| adopt efficient supply chain practices | | | |
| Top management play an effective role in coordination between | | | |
| supply chain partners | | | |
| Top management makes consultations before selecting supply | ì | | |
| chain partners | | | |
| Top management plays an effective role in conflict management | 7 | | |
| on supply chain | 9 | | |
| Top management supports coordination among supply chain | | | |
| partners is high | | | |

SECTION D: SUSTAINABLE SUPPLY CHAIN

With regards to Sustainable Supply Chain, please tick $\lceil \sqrt{\rceil}$ appropriately the number that reflects your firm's present condition. The item scales are five-point Likert type scale with 1 = Extremely Worse, 2 = Worse, 3 = Neutral, 4 = Better, 5 = Extremely Better.

| Sustainable Supply Chain | 1 | 2 | 3 | 4 | 5 |
|--|----|---|---|-----|----|
| Company policies has resulted in significant improvement in terms | | | | | |
| of sales and market share. | 7 | | | | _ |
| Company policies has resulted in significant reduction in terms of | | Ą | | | |
| waste and its disposal costs. | -3 | | Ŧ | | 7 |
| Company policies has resulted in significant improvement in | 7 | X | | 7 | |
| terms of resources management efficiency. | | | 7 | | |
| Company policies has resulted in to respond to and accommodate | 7 | 7 | X | | |
| new products, new markets or new competitors | | | | Ý., | |
| Company policies has resulted in significant improvement in its | × | | | | |
| image in the eyes of its customers. | | | | 1 | |
| Company policies has resulted in significant improvement in | 1 | | | 1 | |
| relations with community stakeholders. | | | | | |
| Company policies has resulted in significant reduction in | | | | | |
| consumption for hazardous/harmful/toxic materials. | | | / | 7 | 2/ |
| Company policies has resulted in significant reduction in energy | | | 1 | 3 | |
| consump <mark>tion.</mark> | | | | 1 | |

SECTION E: SUPPLY CHAIN CHALLENGES

Please indicate your opinions by rating the challenges of Supply Chain Management Practices within your organisation. The rating scale are a five point Likert scale, ranging from 1= not a problem, 2=slight problem, 3=frequent problem but can be managed, 4=major problem and 5= threat to the organization.

| Challenges of Supply Chain Management Practices | 1 | 2 | 3 | 4 | 5 |
|---|---|---|---|---|---|

| Our company lack professional expertise to implement SMC | | | |
|---|---|--|--|
| Our organization lacks frequent Supply Chain Management training | | | |
| Our company has rising logistics cost (Transportation, Material handling and procurement) | | | |
| The company has low level of cooperation among user departments | Т | | |
| Inadequate Supplier relationship with our partners | | | |
| Our company has poor Information flow between partners | | | |
| We do not only concentrate on our core service | | | |
| We lack adequate storage capacities | | | |
| Inadequate measures for monitoring and evaluation of SCM practices to ensure supply chain performance | | | |

| Please indicate measures that can help address the challenges and specify any additional comment or suggestions. |
|--|
| |
| |
| Thank you very much for your time! |
| THE STATE OF THE S |
| WY SANE NO BA |